

Make Your Publications Visible.

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Carline, Nicholas F.; Linn, Scott C.; Yadav, Pradeep K.

Working Paper

Corporate governance and the nature of takeover resistance

CFR Working Paper, No. 14-01

Provided in Cooperation with:

Centre for Financial Research (CFR), University of Cologne

Suggested Citation: Carline, Nicholas F.; Linn, Scott C.; Yadav, Pradeep K. (2014): Corporate governance and the nature of takeover resistance, CFR Working Paper, No. 14-01, University of Cologne, Centre for Financial Research (CFR), Cologne

This Version is available at: https://hdl.handle.net/10419/95924

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



CFR WORKING PAPER NO. 14-01 corporate governance and the wature of takeover resistance N.F. carline • s.c. Linn • P.K. Yadav centre for rinancial research Look deeper

Corporate Governance and the Nature of Takeover Resistance

Nicholas F. Carline

Birmingham Business School, University of Birmingham, Birmingham, B15 2TY, U.K.

Scott C. Linn

Michael F. Price College of Business, University of Oklahoma, Norman, OK 73019, U.S.A.

Pradeep K. Yadav

Michael F. Price College of Business, University of Oklahoma, Norman OK 73019, U.S.A.

Abstract

We investigate the relation between corporate governance characteristics of hostile takeover targets and the choice to employ 'harmful' resistance that is not perceived as being motivated by shareholders' interests. We find that harmful resistance is associated with firms where managers have more pronounced ownership-based and age-related incentives for control, and directors have equity interests less aligned to stockholders. These firms also have less independent boards, are exposed to weaker discipline from outside blockholders, and are inferior performers. In the presence of harmful resistance, the market is less optimistic about the chances of bid completion, and there is a greater likelihood of managerial turnover.

JEL classification: G34

Keywords: Takeover bid; Resistance; Corporate Governance; Stockholder returns; C.E.O. turnover.

This version: March 23, 2014

Pradeep Yadav (corresponding author) is at the Price College of Business, Division of Finance, 307 W. Brooks, Room 205A, Norman, Oklahoma 73019; Phone: (405)-325-5591; Fax: (405)-325-5491; Email: pyadav@ou.edu. The authors thank Eric de Bodt, Dirk Jenter, Kate Holland, Hamed Mahmudi, David Mauer, William Megginson, Thomas Moeller, Michael Rebello, Vahap Uysal, Greg Waller, Athena Zhang, Harold Zhang, and seminar participants at University of Texas at Dallas, University of Oklahoma, Lancaster University, City University of London, SKEMA 9th Corporate Finance Day, and annual meetings of the American Finance Association and the Financial Management Association for valuable comments and suggestions. The authors are also grateful to the Economic and Social Research Council of the U.K. (Grant Reference: RES-000-22-1900) for financial support.

Corporate Governance and the Nature of Takeover Resistance

1. Introduction

Resistance by the management of a takeover target firm can be perceived to be undertaken in the interests of its shareholders, or perceived to be motivated by other considerations, like managerial entrenchment behavior, that potentially neglect or negatively impact shareholders' interests (hereafter 'harmful resistance'). Apart from the seminal contribution of Dann and DeAngelo (1988), the extensive extant literature on hostile takeovers has devoted little attention to the distinction between takeover resistance that is 'adverse' for shareholders, and resistance that is in shareholder interest. In this paper we investigate the factors influencing the choice by target managers to employ harmful resistance following an unsolicited and rejected takeover bid.

We empirically investigate the forces driving target company resistance by modeling the managerial choice of whether to implement harmful resistance. The general consensus in the literature is that corporate governance can influence company value through an effect on manager choices, although the level of the effect is subject to debate (compare for instance Gompers, Ishii and Metrick, 2003, and, Cunat, Gine and Guadalupe, 2012). Our empirical results show that harmful resistance responses are more likely in situations involving weak incentives and weak corporate governance. These results hold after accounting for potential endogeneity issues. Firms in our sample that institute resistance strategies that neglect shareholder interests have Chief Executive Officers (C.E.O.s) with more pronounced ownership-based and age-related incentives for control, and other directors with equity interests less

-

¹ An example of the latter is the 27 billion euros hostile takeover of Arcelor by Mittal Steel in 2006 – one of the largest ever hostile takeover contests in Europe. After a period of staunch shareholder opposition, and in what amounted to the overriding tactic in Arcelor's bid defence against Mittal Steel, Arcelor finally agreed to acquire a large stake in a steel producer Severstal that would give rise to a merger and also allow its top directors to continue to remain in office. See the Financial Times throughout May and June 2006 for a comprehensive commentary on the Arcelor/Mittal Steel takeover contest, especially the reports on May 26, 2006 and June 26, 2006.

² Becht, Bolton and Roell (2005, pg. 16), in an extensive survey of the literature, say: "Little or no work to date has been devoted to the question of identifying which actions or investments constitute 'entrenchment behavior' and which do not."

closely aligned to stockholders. Firms that follow such strategies have less independent boards, are exposed to weaker discipline from active outside blockholders, and are inferior performers.

We also show that such actions do not go unnoticed by the capital market. In new results to the literature, we find the capital market-wide assessment of the probability the bid will be successful is significantly smaller for bids accompanied by harmful resistance. Finally, using the choice to engage in harmful resistance as a potential indicator of a more deep-seated managerial agency problem, we also find that harmful resistance is associated with post-bid consequences for target C.E.O.s. We specifically find that an increased likelihood of subsequent C.E.O. turnover is associated with a bid in which target managers engage in harmful resistance. However, while C.E.O. turnover can be more frequent following such events, we find that the longer-term employment opportunities of target C.E.O.s, including subsequent executive positions as well as director positions, are not directly affected any more than for other resisting firms. Our evidence does show however that the effect of C.E.O. turnover per se is associated with smaller likelihood of longer-term post-bid employment opportunities.

Classifying resistance as either harmful or otherwise must ultimately involve assessing how the capital market views such resistance. The criteria we use to identify harmful resistance relies both on previously documented results on specific defensive resistance actions, as well as capital market price change indicators at the time the resistance is announced. The former criterion is specifically motivated by Dann and DeAngelo (1988) in that we too include all defensive asset restructuring and other blocking tactics that are *prima-facie* suggestive ex-ante of managerial entrenchment, or at least actions not overtly or necessarily motivated by shareholder interests. For the latter criterion, we specifically consider whether the target company's share price reacts negatively to announcement of the resistance, an indicator we take as not being in the best overall interest of shareholders. Such cases could include

actions driven by a managerial entrenchment motive as well as managerial neglect of the 'duty of loyalty'.³

Research examining takeover resistance has generally aggregated all forms of resistance together or focused on a specific type of resistance tactic (see Betton, Eckbo and Thorburn, 2010, for a recent survey of the literature on corporate takeovers.) Examples include the adoption of poison pills or antitakeover provisions (DeAngelo and Rice, 1983; Linn and McConnell, 1983; Jarrell and Poulsen, 1987; Brickley, Lease and Smith, 1988; Agrawal and Mandelker, 1990; Brickley, Coles and Terry, 1994; Comment and Schwert, 1995; Heron and Lie, 2006), the use of defensive cash payouts (Denis, 1990; Heron and Lie, 2006), and the use of staggered boards (Bebchuk, Coates IV and Subramanian, 2002; Bebchuk and Cohen, 2005; Faleye, 2007). In contrast our interest is in disaggregating resistance choices into those that adversely affect shareholders and those that do not. In this way, this paper follows Dann and DeAngelo (1988), who examine a multiplicity of takeover responses by target managers. Our study differs in that we relate the choices made to the internal and external governance characteristics of the target firms involved, and investigate how markets interpret and use information about these choices.

Unsolicited takeover bids play an important role in the market for corporate control. Ex-ante, the threat of a takeover bid represents an external control mechanism serving in principle to motivate managers to operate the firm they oversee efficiently (Fama and Jensen, 1983). Ex-post, successful bids serve to reallocate corporate resources to their highest valued use and provide a means for disciplining errant or incompetent managers. The ex-ante threat of takeover however can be influenced by target managements' ability to thwart the offer once it is made. A positive market response to takeover resistance has generally been interpreted as evidence of resistance acting to enhance shareholder wealth via a better bargaining position and the ability to extract a high premium (Schwert, 2000; Bates and Becher, 2012). On the other hand a negative reaction has been interpreted as evidence of managers

_

³ "The duty of loyalty requires directors and officers to act in good faith, to act in the best interests of the corporation and its stockholders, and to refrain from receiving improper personal benefits as a result of their relationship with the corporation." (Forrester and Ferber, 2012, at http://www.mofo.com/files/Uploads/Images/RRDonnelley-2012-Fiduciary-Duties.pdf). And similarly as pertains to the U.K., see Part 10, Section 172 of the Companies Act (http://www.legislation.gov.uk/ukpga/2006/46/contents).

seeking to cement their entrenchment and extend their private benefits of control (Huang and Walkling, 1987; Cotter and Zenner, 1994).

Target management actions designed to thwart an offer with the intent of entrenching management and/or preserving the inefficient use of corporate resources are at odds with maximizing shareholder wealth (Schwert, 2000; Bebchuk and Cohen, 2005). Alternatively, bid resistance by target management can represent a strategic maneuver designed to maximize the price offered to target shareholders (Stulz, 1988; Berkovitch, Bradley and Khanna, 1989; Berkovitch and Khanna, 1990; Schwert, 2000). The choice of how an unsolicited takeover bid is resisted therefore has both important economic implications as well as potential links to the incentives and pressures faced by target managers, in particular those established by the corporate governance umbrella covering the target firm. In this study we ask and answer the following question: conditional on an unsolicited offer being rejected by a target firm, what determines whether resistance is driven by motives based on shareholder interest; or is driven by considerations like managerial self-interest that neglect, or are adverse to, the interests of shareholders? The U.K. takeover market provides a natural setting for the empirical investigation because U.S.-style, pre-emptive, anti-takeover mechanisms are absent, and thus do not contaminate managerial resistance in direct response to an unsolicited bid. Our results emphasize the relation between the internal and external corporate governance characteristics of targets in unsolicited offers that actively resist, the types of resistance choices selected by target managers, and the implications of those choices.

How managers respond to bids is an important economic factor influencing both the payoffs to target stockholders as well as the distribution of corporate resources (Jensen and Ruback, 1983; Jarrell, Brickley and Netter, 1988; Andrade, Mitchell and Stafford, 2001). The debate over the efficacy of takeover resistance is by no means settled and has led some to propose that resistance should be forbidden (Bebchuk, 2002). Explanations for why managers resist takeover bids are typically couched in terms of stockholder interest motives or managerial self-interest. The 'stockholder interest' hypothesis proposes that managers resist bids because they have superior information about the true value of the firms under their control and seek to extract a takeover premium for target stockholders reflecting this valuation

(Franks and Mayer, 1996; Schwert, 2000). Resistance motivated by stockholder interests would therefore involve strategies that increase information about company fundamentals and any value gap, and/or that would incite competing bids. The opposing hypothesis states that target company managers resist bids to deliberately avoid the potential personal consequences of the firm being taken over. Such consequences might include the loss of any private benefits of control, including any non-tangible utility that comes from overseeing a large organization and any labor market discipline that might follow a successful takeover (Morck, Shleifer and Vishny, 1988). For these reasons our main focus is on unsolicited offers that are in fact resisted.

Resistance by target management is often associated with the notion of the offer being viewed as hostile. Schwert (2000) finds that a commonly used measure of hostility, as identified by the commercial vendor Securities Data Corporation (S.D.C.), does not lead to the conclusion that the valuation impacts of hostile versus friendly offers are different. Furthermore, he finds that strategic bargaining, rather than non-value maximizing behaviour by target management, is the motivation for hostility. Schwert (2000) concludes that bid resistance is a bargaining tool that is used by target management to improve the terms of the takeover for their shareholders. Schwert (2000) treats all hostile offers as a single group whereas in our study, as mentioned earlier, we identify resisting targets on the basis of whether or not they engage in harmful resistance that is not motivated by shareholder interest. Aside from Dann and DeAngelo (1988), studies of takeover resistance have not discriminated between resistance motivated by managerial selfinterest and resistance motivated by shareholder interests. Our study makes such a distinction allowing us to draw new empirical conclusions about the determinants and effects of harmful versus other bona fide resistance. Contemporaneous research by Bates, and Becher (2012) also focuses on takeover resistance. In contrast to their study we specifically differentiate between 'good' and 'bad' resistance, and the determinants and implications of the type of resistance chosen by target managers. In this way we add an alternative perspective on takeover resistance that has largely gone unaddressed in the literature.

In addition to the above, this research also contributes to the segment of the literature focusing generally on the impact of managerial entrenchment or malfeasance on mergers and takeovers. The

conclusion of this branch of the literature has been that managerial entrenchment is associated with bad acquisitions (Jensen, 1986; Masulis, Wang and Xie, 2007; Harford, Humphery-Jenner and Powell, 2012).⁴ While this literature has tended to focus on bidder decisions, our contribution is on the other side, focusing instead on the question of how managerial self-interest or entrenchment influences the choices made by targets when faced with a takeover bid.

Section 2 describes our sample and how we classify target firms in terms of resistance choice. Section 3 describes predictions and the variables we examine. Section 4 reports and discusses results on the relations between the variables described in Section 3 and target company resistance choice. Section 5 examines the relation between the capital market's assessment of the probability of success and managerial resistance choice. Section 6 reports on C.E.O. post-bid turnover and longer-term employment opportunities of target company C.E.O. s. Section 7 presents our conclusions.

2. The sample

2.1 Sample identification

Our empirical investigation is based on the U.K., which provides a natural setting for focusing on target resistance activities following an offer because the ex-ante structural takeover defenses mentioned earlier that potentially deter takeovers are absent. A distinguishing feature of the U.K. market for corporate control is that the regulatory environment in the U.K. effectively shields stockholders from U.S. style shark repellents or poison pills.⁵ This is an important institutional feature in the context of our

_

⁴ Masulis, Wang and Xie (2007) and Harford, Humphery-Jenner and Powell (2012) take the Gompers, Ishii and Metrick (2003) index as an ex-ante proxy for governance where weak governance implies a form of entrenchment and test whether the index is associated with bad acquisition choices. Such an index, even if it is a reliable way of capturing entrenchment, is less applicable in the U.K. because the regulatory setting has always protected stockholders from anti-takeover provisions of one kind or another.

⁵ See Rule 21 of *The Takeover Code*, available at http://www.thetakeoverpanel.org.uk/the-code/download-code.

research question, since it allows us to focus on direct responses to an offer in a setting in which ex-ante anti-takeover devices are absent, and thus do not contaminate our investigation of resistance.⁶

Our sample is drawn from the population of unsolicited takeover bids for U.K. firms during the period 07/1989-12/2003. We selected this sample period for three primary reasons. First, there was limited disclosure of corporate governance structures pre-1989 in the U.K.. Second, for the part of our analysis dealing with the potentially disciplinary consequences of bid resistance, we need to track the aftermarket employment opportunities of target C.E.O.s for up to 3 years beyond the completion of the bid. Third, there were relatively few unsolicited bids immediately preceding and during the global financial crisis.

The initial sample is identified from Acquisitions Monthly and the S.D.C. Platinum Mergers and Acquisition Database and includes completed as well as failed unsolicited takeover bids. We exclude takeover targets that are not publicly traded or are traded in the alternative investment market, those operating in the financial/real estate industry and those operating in heavily regulated industries, specifically the utility/telecommunication, public transport, broadcasting, and newspaper industries.

The latter half (1997-2003) of our sample period was associated with a shift to a more stringent corporate governance regime in the U.K. including increased emphasis on board independence from top management. In particular the 1997-2003 period saw the publication of the "Combined Code: Principles of Good Governance and Code of Best Practice" in 1998 and refinements to the Code, along with the implementation of disclosure requirements on compliance with the Code imposed by the London Stock Exchange (refer to: http://www.ecgi.org/codes/documents/cadbury.pdf). The fraction of unsolicited offers during 1989-1996 was roughly 18.3% of the total offers for that period while the number was 10.2% for the period 1997-2003. One explanation for the decline in the proportion of unsolicited offers is

-

⁶ See Stout (2002) for a discussion of the potential ex-ante benefits of anti-takeover provisions. Aside from Rule 21, the Companies Act provides that investors in U.K. companies are also entitled to call for an Extraordinary General Meeting if the shareholders calling for this meeting hold 10% or more of the voting shares. At such a meeting those shareholders may put forward resolutions to appoint and/or remove directors. In other words, even though most directors in the U.K. are only put up for election once every 3 years (staggered board), a would-be acquiring company – provided it has sufficient shareholder support – can oust the management of a U.K. listed company in a relatively short space of time.

that the increased attention to stockholder interests due to the shift in governance emphasis may have resulted in a greater willingness on the part of management to negotiate friendly deals consistent with the finding in Bange and Mazzeo (2004) that more independent boards are more likely to negotiate. While we do not emphasize these results, consistent with the presence of a time effect we observe significantly higher percentages of independent directors and outside blockholdings during the 1997-2003 period for our sample together with a greater incidence of bidders approaching targets prior to the bid and harmful resistance. As such we control for this time effect in the analyses that follow.

2.2 Resistance identification and classification

We identify resistance activities implemented by target companies by reading the full text of every offer-related announcement filed with the Regulatory News Service (R.N.S.) of the London Stock Exchange (L.S.E.) for each bid in the sample. We define a publicly resisted (rejected) bid as the first takeover attempt of a target firm to be formally initiated and then rejected within a period of at least 1 year. We account for all subsequent offers, including those from third parties, officially announced up to 1 year after the resolution of the first bid. The final sample includes 121 publicly resisted takeover bids for 119 unique target firms. The only target firms appearing twice had sample bids separated by at least 2 years. The sample therefore includes only unsolicited offers that are resisted, and, as such, the results we present are conditional on both the joint observation of an unsolicited bid, and resistance to that bid by the target, allowing us to focus on the motives for resistance. To the extent that we are unable to account for any firms that might otherwise have resorted to harmful resistance had they not either been immune to takeover bids or been able to negotiate acceptable offer terms, our results will be biased against finding any association between corporate governance and takeover resistance. That would also be the case for any firms that might otherwise have engaged in *bona fide* resistance had they not been able to negotiate acceptable offer terms.

Arguably, certain specific types of defensive restructuring resistance actions are *prima-facie* suggestive ex-ante of managerial entrenchment, or at least actions not overtly or necessarily motivated by shareholder interests. These include: (1) Spin-offs/sell-offs which are divestments that deny bidders

access to assets of value, (2) Mergers/acquisitions/joint ventures which make target firms more cumbersome to acquire due to size, strategic, or antitrust grounds, or, which eliminate bidders directly, (3) Stock repurchases/special dividends which involve payouts that eliminate a bidder's access to excess cash held by the target, and, in the case of targeted repurchases, that increase the proportion of stock in the hands of friendly (to management) stockholders, (4) Going private transactions which use competing management buyouts to create bidding contests, potentially reducing bidder expected gains, (5) Management changes which make removal of newly appointed officers especially costly because of special contractual payments triggered by a takeover, and (6) The enlistment of friendly third party participants who acquire stakes that will effectively block the takeover leaving the extant management team in control.⁷ Dann and DeAngelo (1988), in a study of resisted U.S. takeover contests, show empirically that share prices fall, on average, in response to the above-listed types of defensive restructuring resistance activities, and conclude that such activities are evidence of managers seeking to cement their entrenchment. On the other hand, resistance actions that, for example, resolve information asymmetry by enhancing the availability of value relevant information about the higher true worth of target firms (relative to the bid), strengthen the relative bargaining positions of shareholders during the course of takeover bids, and are in the interests of shareholders.⁸

We use the six specific types of defensive restructuring resistance actions listed above as our starting point for classifying cases in our sample that represent harmful resistance. We then identify those cases in this subgroup for which there was a negative abnormal stock price reaction associated with the resistance announcement. We use market adjusted daily returns to estimate the reactions where the market proxy is the F.T.S.E. All Share Index. Our reliance on the negative capital market reaction to the resistance announcement provides a market-driven classification method based upon how shareholders greeted the resistance, given that a negative reaction would in general be inconsistent with the action

-

⁷ The potential ramifications of many of these activities are reviewed in the extensive survey of Betton, Eckbo and Thorburn (2010).

⁸ Examples are releasing new financial and strategic information measures to aid in justifying and communicating higher target firm valuations, lobbying significant stockholders, and soliciting third party takeover attempts.

being perceived as in the best interests of shareholders.⁹ There are fifty cases in the harmful resistance group so identified, 41% of the total sample of 121 unsolicited and resisted offers. If the classification system we employ is excessively noisy it should reduce the power of the tests we report to reject the null.

Frequencies of the specific resistance tactics employed by the 50 targets in the harmful resistance group are presented in Table 1. Spin-offs/sell-offs are the most commonly employed measures, with defensive divestment proposals being used by 46% of the firms in the focus group. Mergers/acquisitions/joint ventures are used in 30% of the cases, and 16% employ stock repurchases/ special dividends. Going private transactions, management changes, and the use of white squires are used less frequently, in 10%, 10% and 8% of the cases respectively. On average, 1.2 different tactics are used by target boards in the harmful resistance group. The proportion of targets using spin-offs/sell-offs drops from roughly 58 % during the 1989-1996 period to 35% during the 1997-2003 period. In contrast, there is nearly a trebling in the percentage using stock repurchases/special dividends. Table 1 also displays the average abnormal returns for each tactic group at the time the action is announced. We report averages only for brevity, but emphasize, as is displayed in column 3 of the table, that the percentage of cases with a negative reaction is equal to 100% for each category.

Roughly 15% of all 121 targets in the sample engaged in asset restructuring prior to a bid and roughly 21% had management changes. The fractions did not however differ between the first and second halves of the sample period (1989-1996 and 1997-2003) (Z-statistics for differences in proportions: 0.15 (asset restructuring) and 0.06 (management changes)). These results suggest the resistance strategies used by the sample firms were not likely the result of changes in the tendencies of boards to implement such strategies prior to a bid.

Thirty four percent (34%) of the target boards (out of 121) directly solicit third party bids. The fraction is significantly higher (abs. Z-statistic: 2.91) during the latter half of the sample period (49% versus 24%). Thirty nine percent (39%) of these cases involve members of the harmful resistance group

_

⁹ That said, for our sample, the overwhelming majority of cases - almost all - falling within the ambit of the six defensive resistance strategies listed above as having the potential to be adverse to shareholder interests, display resistance-announcement related negative abnormal returns.

and as such represent defenses of last resort (or 'white knights') as distinct from the remaining cases of pure bidder solicitation. Furthermore, where a 'white knight' ends up acquiring the target, the C.E.O. remains in place 50% of the time compared to only 28% for cases of pure bidder solicitation. Twenty four percent (24%) of the sample firms issued higher (defined as at least 10%) profit announcements when opposing takeover offers. This proportion declined significantly (absolute Z-statistic: 2.20) between the first and second halves of the sample period (14% versus 31%). Twenty four percent (24%) of these cases involve members of the harmful resistance group indicating that these actions, in contrast to those defining that group, are more consistent with target managers seeking improved payoffs for shareholders.

3. Predictions and variable measurement

3.1 Internal and external incentives and governance and resistance choice

A target's choice to resist may be influenced by the internal and/or external governance conditions it faces at the time of the bid. For instance Berkovitch and Khanna (1990) develop a theoretical model in which target managers' underlying motivation is shareholder wealth maximization but in which defensive strategies arise and are optimal. However, the authors also conclude that resistance strategies can lead to negative revaluations of the target. This can occur when managers use resistance to entrench their positions. The effect is compounded when the effectiveness of managerial monitoring by shareholders is weak (Berkovitch and Khanna, 1990, section III.B.). Mechanisms that may link shareholder and manager interests include such things as the alignment of manager wealth and income with stockholder wealth, the configuration of the board of directors, and the presence of investors who hold large blocks of shares giving them the potential to influence manager actions.

Some empirical evidence suggests target boards are less inclined to resist bids per se the greater are the potential net wealth gains for target managers in the event of a takeover (Walkling and Long, 1984; Cotter and Zenner, 1994). Cotter and Zenner (1994) conclude that the fraction and value of managerial ownership is the driving force and that these personal wealth effects exceed the present value of lost compensation resulting from dismissal following a successful takeover. Demsetz (1983), Fama

and Jensen (1983), Stulz (1988) and Morck, Shleifer and Vishny (1988) have argued that large managerial ownership stakes can give rise to managers extracting greater private benefits of control if those stakes allow them to maintain control. If the personal benefits to managers from entrenchment outweigh the losses they incur from not pursuing shareholder wealth maximization then there is an incentive for them to seek to maintain the status quo. Larger ownership stakes could therefore promote resistance for entrenchment reasons and might in fact be selected strategically by target managers. We use the percentage of outstanding stock held beneficially by the C.E.O. of the target company to test for the presence of these effects. Information on ownership stakes for our sample companies along with the additional variables described in this section and those that follow are obtained from several sources: Companies House filings, the Corporate Register, the R.N.S. and Datastream. ¹⁰

Holmström (1982) and Gibbons and Murphy (1992) argue that agency problems worsen as managers approach retirement and begin to care less about their long-term career prospects. This suggests that a positive relation may exist between C.E.O. age and resistance driven by managerial self-interest. In the same vein, Dechow and Sloan (1991), Antia, Pantzalis, and Park (2010) and Serfling (2012) provide evidence that older managers focus excessively on actions with short-term gains. Jenter and Lewellen (2011) provide evidence that C.E.O.s closer to retirement are more willing to accept takeovers at lower premiums, that is they may be less prone to resist. Taken together we capture these effects by hypothesizing and testing the proposition that C.E.O. age is non-linearly related to resistance with age becoming a less important factor as retirement age is approached. We capture this effect using C.E.O. age and its square.

Gillan (2006) refers to the board of directors as "... the lynchpin of corporate governance" (p. 385). This view stems from the fiduciary obligation of the board to shareholders, and the board's principal role of monitoring and disciplining management (see for instance, Bhagat and Black, 1999,

-

¹⁰ All U.K. registered companies have always been required to deposit a copy of their annual report and accounts, and other director and stockholder related filings, with Companies House. The Corporate Register has been published at least bi-annually since 1989 and provides a means of tracking executive and director positions across U.K. listed companies. The R.N.S. has always collated all material news in an as disclosed form from listed companies to the L.S.E.. Stock prices and returns and financial data are sourced from Datastream.

2002; Hermalin and Weisbach, 2003). The beneficial ownership of directors other than the C.E.O. has the potential to enhance director incentives to monitor managerial behavior. At the same time, directors other than the C.E.O are less likely to enjoy private benefits of control. Boards that are aligned with shareholder interests may in principle seek to keep harmful resistance actions of managers in check. Under such conditions we would expect that more aligned boards would be less likely to support harmful resistance. The percentage of the outstanding equity shares beneficially held by directors other than the C.E.O. is our measure of the direct alignment between those board members and stockholders.

Directors may also be more prone to serve the interests of stockholders if the board is sufficiently independent of the C.E.O. Directors are broadly deemed to be autonomous when no discernible personal, employment, or business connections to the firms on whose boards they reside exist nor to the managers of those firms. Directors who serve on the boards of more than one firm have added incentives to act as good stewards because this contributes to the building up of their reputation capital in the market for directors. One force potentially driving independent directors to act in the interest of shareholders is the value of their reputation in the director labor market (Kaplan and Reishus, 1990; Shivdasani, 1993, in the specific context of takeovers). We construct two alternative measures of target board independence. The first measure is based on board composition and equals the percentage of the board members who are outsiders and also hold at least one, non-interlocking, directorship at another firm. A non-interlocking directorship is defined as one in which the non-aligned director can serve as a C.E.O., but on whose board the target C.E.O. does not reside. Our second measure is a dummy variable that takes the value 1 if an outside director holding at least one, non-interlocking, directorship at another firm is also chair of the board and 0 otherwise. If independent boards act in shareholder interests we expect the more independent are resisting target boards the less likely they are predicted to engage in harmful opposition tactics.

Another important characteristic of the board that may play a role in board decision-making is the number of board members. Yermack (1996) has shown that large boards tend to be associated with lower

_

¹¹ In which the eligible companies in question are members of a non-regulated industry (not in the financial/real estate, utility/telecommunication, public transport, broadcasting, and newspaper industries) and have a full listing on the L.S.E..

firm valuations. Jensen (1993) argues that boards containing more than 8 members function less effectively and are easier for the C.E.O. to control. In contrast Coles, Daniel and Naveen (2008) suggest a non-linear relation exists between board size and value. We measure the size of the board using the number of target firm directors.

Outside blockholders can potentially play an active role in corporate control (Shleifer and Vishny, 1986) and can facilitate hostile takeover bids (Shivdasani, 1993). Target managers' responses to an unsolicited bid may be influenced by the pressure brought to bear on them by outside blockholders. We define a non-aligned blockholding as a blockholding of at least 5% of total equity votes, over which managers do not exercise control and that is not held by any target director. We use total shares held by such blockholders as a percentage of all shares to measure aggregate blockholder influence. Initial bidder toehold interests are excluded from the measure of blockholder ownership.

All data on governance related variables are measured as close as possible to, but before the bid date.

3.2 Other factors

Franks and Mayer (1996) and Schwert (2000), among others, find that unsolicited offers are not necessarily directed at only inefficiently managed firms. However, the managers of inefficiently managed target firms may stand to lose more (jobs, reputations, private benefits of control, etc.) than their counterparts at firms that are efficiently managed as a result of a successful takeover. If true such managers are predicted to engage in entrenchment motivated resistance. We measure company efficiency using the pre-bid market to book value of the target firm and the target's asset turnover ratio both for up to 2 years ending at the fiscal year-end prior to the bid. Market value of assets is computed as the common stock price multiplied by the number of outstanding shares, plus the book value of non-equity securities and total debt. The stock price is measured 30 days prior to the bid or prior to any official rumor period, whichever is longer. The book value of tangible assets (total assets minus intangibles) is used as the denominator for both performance measures.

The cost of acquiring information about target firms has been inversely linked to managers' decisions to resist takeover bids (Fishman, 1988, for a theoretical justification and Jennings and Mazzeo, 1993, for empirical evidence). Managers of target firms with relatively more publicly available information who choose to resist are likely not motivated by a desire to increase the information set available about their firm. Our proxy for information availability is based upon firm age. We define firms with less information as those whose shares have been fully listed for 6 years or less (Jennings and Mazzeo, 1993). A binary variable labeled 'Young firm' is used to identify such cases. The natural logarithm of firm size proxied by pseudo market value (as previously defined) is has been employed by numerous authors including Mikkelson and Partch (1989), Cotter and Zenner (1994), and Schwert (2000).

The offer premium reveals in part the bidder's beliefs about the value creation potential of acquiring the target as well as the likelihood that the offer will be met with resistance. In the takeover models of Giammarino and Heinkel (1986), Fishman (1988) and Hirshleifer and Png (1989) for instance a large offer premium reveals a high bidder valuation of the target. Inefficiently managed targets present an opportunity for value creation that may include reorganization and dismissal of the target's top management team as first proposed by Manne (1965). A larger premium may thus be viewed by the inefficient manager as an indicator that if the takeover is successful dismissal may be likely and thus lead to harmful resistance. In contrast, managers who are motivated by stockholder wealth interests (efficient managers) will resort to resistance to improve the terms of what they view as inferior bids. Initial bid premiums have been associated with the likelihood of resistance, with lower premiums, on average, being rejected at a greater rate (Jennings and Mazzeo, 1993). However, such studies do not directly consider what we refer to as harmful resistance. Initial bid premiums are computed as the percentage difference between the offer price and the target firm's stock price, adjusted for the effects of takeover rumor. 12

_

¹² For the case of offers involving an exchange of common stock, the bidder's stock price is taken 5 trading days before the bid announcement date. The bid rumor adjustment is similar to that described for the target firm's market value of assets. Bates and Becher (2012) primarily focus on an abnormal premium (actual minus predicted premium). When we compute predicted premiums in a similar manner from within our sample and include the abnormal premium in place of the actual premium the results to follow are unchanged.

Bidder toe-holds have been shown to be an important element in takeovers and to exhibit both variation across bidders as well as the observation that many bidders hold smaller pre-bid share positions than the maximally legal ownership level (Jennings and Mazzeo, 1993; Betton and Eckbo, 2000; Betton, Eckbo and Thorburn, 2009). Goldman and Qian (2005) develop a model in which optimal toehold positions account for post-bid target value if an offer fails due to resistance, and show toehold positions are endogenous in the model. This occurs within a setting in which the motive for target resistance is to preserve target managers' private benefits of control and the ex-ante probability of takeover is increasing in the bidder toehold. Betton Eckbo and Thorburn (2009) conclude that if the target management is expected to resist regardless of the size of the toehold, then acquiring a toehold is always optimal. This also suggests the toehold may be endogenous. Walking (1985), Jennings and Mazzeo (1993) and Betton and Eckbo (2000) find that toehold bidding increases the probability of takeover success. Managers that choose to resist takeover bids will find it increasingly difficult to do so for entrenchment reasons the greater is the voting power behind these toehold interests. We measure toeholds as the percentage of total equity votes controlled by initial bidders immediately prior to the announcement of the offer. As an alternative proxy for the bargaining power of initial bidders an indicator variable is used to identify cases where either the bidder already has a representative on the target board, or a dissident target director pledges support for the bidder.

Resistance may also be related to the terms of the offer and to whether the bidder makes a genuine approach to the target's board before launching the bid. For instance, the model developed by Fishman (1988) suggests that cash bids are more likely to be accepted. Bids for cash have been weakly associated with less bid resistance (Jennings and Mazzeo, 1993), while bidder approach has been strongly connected to a lower propensity for opposition (St-Pierre, Gagnon and Saint-Pierre, 1996). 'Full cash terms' is a dummy variable that takes the value of 1 if the initial bid is a cash offer and 0 otherwise. 'Bidder approach' is an indicator variable used to identify cases where the initial bidder makes an approach to the target board before launching their bid.

3.3 Descriptive statistics

Table 2 presents descriptive statistics for the target firms. Data are presented for the harmful resistance group and for all other resisting targets. C.E.O.s of target firms classified in the harmful resistance group own an unconditional mean beneficial equity interest of 2.30% of their firms' stock, although the median stake is considerably less at only 0.09%. In contrast mean C.E.O. holdings for the remaining targets equals 1.47%. The unconditional means (and medians) of the two groups are not significantly different using a mean difference t-test or the Kruskal-Wallis test for equality of medians. The average and median age of a C.E.O. in the sample is 52 years. The difference in the unconditional means (medians) of C.E.O. age between the two groups is not significant.

Directors of target firms in the harmful resistance group, excluding C.E.O.s, have a mean and median personal stock interest of 1.24% and 0.30%, respectively. In contrast director ownership for the remaining targets is larger: mean 2.89%, median 0.44%. However conventional tests indicate neither the unconditional means nor medians are statistically different between the groups. C.E.O. and other director equity stakes do however exhibit cross-sectional variation (harmful resistance group: 6.3% and 2.5%; all other targets: 4.1% and 8.2%).

Targets in which the C.E.O. and board chair positions are separated and where the board chair is also an independent director are observed less frequently for firms in the harmful resistance group (20% versus 45%) and the difference in the proportions is statistically significant. Independent directors constitute roughly 17% of the board for the harmful resistance group and about 19% for all other targets. The mean (and median) number of directors is roughly the same for both the harmful resistance group and all others and equals 7 members. Outside blockholdings are statistically significantly smaller for the harmful resistance group (25% versus 32%) and this is true for both the means and the medians.

The mean (median) offer premiums are not statistically different between the two groups. This is also true for the fraction of the target's shares held as a toehold, for the fraction of cases in which a dissident director sympathetic to the bidder was present, for the fraction of cases in which a full cash offer was made, and whether the bidder approached the target prior to the bid.

Average and median statistics for the company performance proxies reveal that, for the 2 year period preceding the initial takeover bids, non-harmful resisting firms have market to book value ratios in excess of the harmful resistance group and have asset turnover ratios that are also larger. But these differences are not statistically different from zero. The non-harmful resistance group is made up of younger firms (those that have been fully listed 6 years or less) and the difference in the fractions of younger firms in both groups is statistically significant.

There is a 1.6 fold increase between the periods 1989-1996 and 1997-2003 in the mean number of independent directors on the boards of resisting targets. This result is consistent with the overall structure of U.K. boards having changed significantly in response to numerous independent reports on corporate governance practices and the publication of the "Combined Code: Principles of Good Governance and Code of Best Practice". This finding also potentially explains our earlier observation that a relatively smaller percentage of takeover offers are unsolicited during the period 1997-2003. That is, a greater degree of board independence may have eased the path for would-be acquirers to privately negotiate deals with incumbent directors. We also find that the incidence of bidders approaching targets before a bid increased from 19% to 43% between the sub-periods although curiously a greater fraction of such approaches is observed for cases involving targets that ultimately follow harmful resistance strategies. This latter result is perhaps consistent with bidders seeking out firms that are not performing as well only to be dismissed by target management. Taken together these observations are consistent in spirit to those of Boone and Mulherin (2007) who find a takeover market in the 1990s that is increasingly characterized by pre-public negotiation.

4. Resistance, managerial incentives and corporate governance

Table 3 presents the average marginal effects of variables selected to represent the hypotheses described in Section 3 on the probability of choosing a value decreasing resistance strategy (harmful resistance). We model the choice variable as a probit specification. Similar results (not tabulated) are found for logit specifications. Marginal effects in probit (or logit) models are related to all of the

estimated coefficients of the model and hence examining the estimated coefficients by themselves can lead to potential misinterpretations (Woolridge, 2002; Long and Freese, 2006). Following convention, marginal effects for binary explanatory variables are measured as the difference in the evaluated probability when the variable takes the value 1 minus the value when it equals 0. The dependent variable takes the value 1 for each of the 50 cases we classify as being associated with value decreasing resistance and 0 otherwise. As already mentioned, our results are conditional on the bid being unsolicited and being resisted. For convenience we refer later to the 'probability' of harmful resistance resistance but it should be understood that what we are speaking of is conditional on the bid being unsolicited and resisted.

It is possible that one or more of the right hand side variables in the model are endogenous. Specifically the explanatory variables offer premium and bidder toehold might potentially be endogenous (Fishman, 1988; Hirshleifer and Thakor, 1994; Cotter, Shivdasani and Zenner, 1997; Bange and Mazzeo, 2004; Betton, Eckbo and Thorburn, 2009). Likewise, C.E.O. and outside blockholder ownership might also reflect strategic choices by managers and investors that could be related to benefits of control and hence to the choice of resistance. In the discussion that follows we comment on the results of tests for exogeneity for each of these variables. We do not present all the results for brevity but in Table 3 do include the results for models that treat the offer premium as potentially endogenous. We present two sets of results in Table 3. In one set of results the offer premium is treated as an exogenous variable by assumption. In the second set of results we treat the offer premium as if it is potentially endogenous and test the null hypothesis that the variable is exogenous using a Wald test (Woolridge, 2002).

Columns labeled 'Probit' in Table 3 report average marginal effects for each explanatory variable in the model treating the offer premium as exogenous by assumption. Z-test statistics for the null hypothesis that the marginal effects equal zero are reported in parentheses. A marginal effect measures the partial of the expected probability of the choice to select a value decreasing resistance tactic with respect to an explanatory variable. The average marginal effect is computed by evaluating the marginal effects at every observation and then averaging (Woolridge, 2002). Columns labeled '2-stage' report

average marginal effects for models in which the offer premium is treated as an endogenous variable. The modeling assumption is that the choice variable is described by a probit model while the continuous endogenous variable is described by a linear model with normally distributed errors. These models are estimated using maximum likelihood methods (Woolridge, 2002). Wald test statistics of the null hypothesis that the variable labeled 'offer premium' is exogenous are reported at the bottom of each column labeled 2-stage. In every case we cannot reject the null hypothesis of exogeneity of the offer premium (p-values for the test all exceed 0.05 by a healthy margin). The resulting estimated coefficients for all other variables are of the same sign and have the same general magnitudes and significance levels as those shown in the columns labeled Probit. The same applies to tests of exogeneity for the other potentially suspect right hand side variables: C.E.O. ownership, outside blockholder ownership and the bidder toehold level. Based upon these collective results we therefore focus our discussion on the results for the basic probit models reported in the columns labeled 'Probit'.

We also include in the models estimated the pre-bid run up of the target's stock, computed as the daily cumulative market (F.T.S.E. All Share index) adjusted return over the 12 month period prior to the bid or prior to any official rumor period, whichever is longer. Industry controls are also included but are not tabulated for brevity. Due to the sample size we identify each target with one of four industry sectors: natural resources, industrials, consumer goods, and services. Dummy variables for industry sector membership are then constructed. The dummy variables for industrials, consumer goods, and services are included in the model with the intercept absorbing the natural resources sector.

¹³ See Newey (1987) and Woolridge (2002). We use the pre-bid run-up of the target's stock and the natural logarithm of target firm size as instrumental variables (see Betton,Eckbo and Thorburn (2010)).

¹⁴ For the maximum likelihood estimation with a single continuous endogenous variable, the test is simply a Wald test that the correlation between the error terms in the structural equation and the reduced-form equation for the endogenous variable is equal to zero.

¹⁵ Following Demsetz and Lehn (1985) and Holderness (2009) we use the natural logarithm of target firm size and the standard deviation of daily target stock returns as instruments for these other potentially endogenous explanatory variables.

The probability of observing a resistance choice placing the target in the harmful resistance group increases with the size of the C.E.O.'s equity ownership interest.¹⁶ This result is consistent with the proposition that larger levels of managerial ownership can give rise to managers wielding more power and extracting greater private benefits of control as suggested by Demsetz (1983), Fama and Jensen (1983), Morck, Shleiffer and Vishny (1988) and Stulz (1988). This observation is plausible given the nature of our sample. Unsolicited and resisted offers suggest that a negotiated deal may have already failed, or that the target's management rejected any overtures from the bidder, both of which could be due to managers seeking to maintain their private benefits of control. Broadly consistent with our findings are the results presented by Moeller (2005) who finds that higher C.E.O. stock ownership, in addition to other proxies for low target stockholder control, is associated with a smaller takeover premium. explanation for the result is that instead of negotiating a large takeover premium that would benefit shareholders, target C.E.O.s reach a side-deal with the bidder insuring they continue to receive private benefits which could come in the form of a generous severance package. Our results differ however from Walkling and Long (1984) and Cotter and Zenner (1994) who find that managers are more likely to reject bids the lower is their ownership stake. Our sample of managerial choices is different from these studies however in that we distinguish between harmful and non-harmful resistance which means we are not pooling all resistance actions together.

Takeover bid resistance is non-linearly related to C.E.O. age. Older managers use harmful opposition strategies with greater frequency, but this effect is partially mitigated for much older C.E.O.s. Neither the C.E.O. stock ownership nor age effects are affected by controlling for top managers who have only recently (within the 12 month period prior to bid announcement) become the top managers of target firms. Our findings for C.E.O. stock ownership and age echo those of Harford (2003) who documents that target C.E.O.s with a higher predicted probability of not being retained have a greater propensity to

.

¹⁶ This result is not sensitive to using the logistic transformation of the percentage of shares held by the C.E.O. (not tabulated) employed in Demsetz and Lehn (1985) and Demsetz and Villonga (2001).

resist, rather than recommend, takeover bids. Later we show that harmful resistance is associated with a higher probability of target C.E.O. turnover.

The results consistently show that boards, excluding the C.E.O., that own more stock are less likely to endorse a harmful opposition strategy. ¹⁷ Results in the literature show that resisting boards tend to have lower ownership interests relative to those favoring takeover bids (Morck, Shleifer and Vishny, 1988; Cotter and Zenner, 1994). Our results indicate that the level of board ownership discriminates between resistance driven by manager-driven motives in contrast to stockholder interests.

The estimation results for models (1) and (3) show that choices that place the target in the harmful resistance group are less probable when the board chair is independent and holds at least one other, non-interlocking, directorship. This is the result we would expect if these chairpersons act as an important counterweight to the C.E.O. and whose reputations are potentially valuable. The result is robust to the inclusion of a control identifying those firms that only initiated a dual leadership structure in the 12 months prior to being the target of a takeover.

In model (5) the board leadership variable is replaced with the percentage of the board that are independent and that hold at least one other, non-interlocking, directorship. The coefficient on this variable while negative is not statistically significant at conventional levels. A significant shift in the corporate governance environment occurred in the U.K. during 1997-2003 including movement towards more outside director representation on boards. We construct an interaction variable equal to the product of 'Independent directors (%)' and a dummy variable that takes the value 1 if the takeover bid occurred during the period 1989-1996 and (0) otherwise. The coefficient of the interaction variable as reported under model (7) is negative and statistically significant. The result indicates that board composition is an important factor but that its influence is restricted to the period 1989-96.

There is no statistically significant association between target board size and harmful resistance.

Using the natural logarithm of board size (not tabulated) does not alter this conclusion. Bange and

¹⁷ In contrast, Harford (2003) documents the potentially adverse impact of takeover bids on non-C.E.O.-directors' (including outside directors') wealth and future board seats.

Mazzeo (2004) find that targets with larger boards are more likely to receive unsolicited offers. All the offers in our sample are unsolicited. Our result indicates that there is no marginal board size effect on resistance choice given an unsolicited offer.

The marginal effect of the variable non-director-aligned blockholdings ('outside blockholdings') on the probability of resistance choices placing the firm in the harmful resistance group is negative and statistically significant. This finding is robust to alternative measures of blockholdings, including a Herfindahl-type measure capturing block holding concentration (not tabulated). The result is consistent with the hypothesis that the presence of blockholders discourages harmful resistance as suggested by Shleifer and Vishny (1986)

The negative effects of the market to book and asset turnover ratios suggest that more efficient managers are less likely to resort to harmful resistance strategies and consequently less efficient managers more likely. This relation is what we would predict if poorly performing managers fear disciplinary consequences and/or engage in the use of corporate resources that do not maximize value. The results are robust to alternative time periods over which market to book and revenue to tangible assets ratios are computed. The significant performance differences observed in Table 3 remain after netting out median industry effects (Morck, Shleifer and Vishny, 1988).

Morck, Shleifer and Vishny (1988) conclude that there is a greater propensity for poorer performing and older firms to be the targets of disciplinary takeover bids. Also, consistent with this view, the results presented in Table 3 reveal that harmful resistance resistance is less likely for younger firms. Information about such firms is also likely to be less precise. Fishman (1988) argues that poor quality/more costly information about target firms creates a bargaining imbalance, which we argue is more likely to induce the use of pro-stockholder resistance strategies. Neither the differential effects related to past performance nor those emanating from the cost of information acquisition are affected by the inclusion of target firm size.

The positive sign for the effect of the offer premium implies a greater likelihood of observing harmful opposition in response to higher initial bid premiums. One explanation for this result is that a

higher premium signals to target managers that the bidder is not only very serious about gaining control but also believes the target's value can be enhanced, possibly through a reorganization of the target's management team. In contrast, managers resorting to harmful opposition resist irrespective of the level of the premium. In contrast, the other targets have chosen to resist because the managers perceive that the initial offer premium undervalues their firm. Varying the target stock price used to estimate initial bid premiums does not lead to different inferences.

Model (1) contains a control for bidder toehold. The results suggest that toehold interests are not associated with the use of harmful resistance strategies. However, we find that bidder toehold and outside blockholdings are highly positively correlated (43%) and the result is statistically significant at the 0.05 level. We estimate the model including bidder toehold but excluding outside blockholdings (not tabulated) and find that bidder toehold has a negative and statistically significant marginal effect. We also substitute the factor identifying cases in the sample where initial bidders have representation on target boards, or are supported by dissident incumbent directors, for the toehold interest variable. The result is a generally significant, negative effect for the variable. Cotter, Shivdasani and Zenner (1997) find that such a factor is significantly and inversely related to bid resistance in U.S. contests. Initial bidders that have representatives, or the backing of dissidents, on target boards are less likely, therefore, to face harmful resistance.

Finally, a cash only offer is not associated with the nature of bid resistance. This is also true for cases in which the bidder approached the target before the public offer. If target managers are intent on retaining their positions and control of the firm then they will resist whether the bidder has made takeover ovations prior to the public bid or not or whether the offer is for cash despite the potential positive aspects of a cash offer (see, Fishman, 1989; and the empirical results of Jennings and Mazzeo, 1993). We treat the choice to offer cash as exogenous. Theory and by and large all empirical studies of the choice to offer cash find that the choice is a function of bidder characteristics in contrast to target characteristics or behavior (Martin, 1996; Faccio and Masulis, 2005).

In summary, takeover bid resistance motivated by harmful resistance is more likely the greater are the personal stock interests of C.E.O.s, and as these top managers approach the peak of their executive careers (grow older). Boards of directors that are more closely aligned with stockholders through equity ownership and/or are more independent of management are associated with a reduced propensity for harmful opposition. Harmful resistance is also a less probable response the more concentrated are outside blockholdings, and the better performing are target firms. Lower information quality about the target decreases the probability of harmful resistance while higher initial bid premiums increase its likelihood.

Overall we consistently reject the null hypothesis that all the coefficients of each of the estimated models are equal to zero using a Wald test.

5. Conditional expectations about the probability of bid success

The results presented in Table 3 suggest that harmful defensive resistance is associated with weaknesses in managerial incentives and governance. If the relations between resistance choice and incentives were simply a statistical artifact we would not expect the capital market to reflect the implications of such resistance in target stock price revisions at the time of the bid. Simply looking at the market reaction by itself is insufficient to tease out the answer to this question. One must account for the expected probability of success as well as the premium offered to assess whether capital market participants initially regard resistance choice as having a direct consequence. We now provide such an examination.

The capital market revaluation of the target at the time of the bid reflects both the offer premium as well as the market-wide assessment of the probability the bid will be successful. In simple terms $\Delta P = \left(Offer\ Pr\ emium\right)x\left(p\right) \text{ where } p \text{ represents the assessed probability the offer will succeed and } \Delta P \text{ represents the change in the target's share price when the offer is for all target shares, and the$

expected price if the offer fails equals the pre-offer price. Of course in an empirical analysis ΔP must be suitably adjusted for general market movements. A natural question is whether the probability of success assessed by the market at the time of the bid is different for firms in the harmful resistance group.

We measure target share price reaction to a bid as the cumulative abnormal return (C.A.R.) over the period from 30 trading days prior to any bid rumor, or the formal announcement date if no rumor circulated, through the offer announcement date. The market index used as a basis for the simple market model is the F.T.S.E. All Share Index. The market model is estimated for each case using the 120 trading days leading up to 30 days prior to the first announcement or rumor date. The mean and median target company C.A.R.s for the announcement interval equal 24% and 26% respectively and both are significantly different from zero at conventional levels of significance, consistent with most results reported in the literature (Franks and Harris, 1989; Betton, Eckbo and Thorburn, 2010).

We estimate the assessed probability of success by regressing target company C.A.R.s on the initial offer premium and several control variables that have been shown to be related to the response of target company share prices to bids (see for instance Betton, Eckbo and Thorburn, 2010). Table 4 reports the results. We are interested in the estimated coefficient on the offer premium which reflects as noted earlier the market assessment of the probability of success. We present estimates for the harmful resistance group of companies and separately for the remaining sample bids in columns (1) and (2) respectively, whereas column (3) reports results of a full interaction model based on a dummy variable defined by membership in the harmful resistance group. The estimated coefficients on the offer premium variable equal 0.422 for the harmful resistance group and 0.654 for all other targets. These numbers represent the assessed probability of success for each group respectively. Clearly the point estimates differ and suggest that the assessed probability of success is smaller (about 42%) for the harmful resistance subsample but nevertheless significantly different from zero. In column (3) we present a full

¹⁸ See Cotter and Zenner (1994). See also Asquith (1983), Bradley, Desai and Kim (1983), Ruback (1988), and Fabozzi, Ferri, Fabozzi and Tucker (1988) for evidence that target company share prices in unsuccessful offers revert back to pre-offer levels.

interaction model in which the interaction dummy equals 1 if the case belongs to the harmful resistance group and 0 otherwise. The coefficient on the interaction variable defined as the dummy times the offer premium is negative (-0.232) and statistically significant at the 0.05 level. The inference is therefore that the market's assessment of the probability of success is significantly smaller for those cases in which resistance reflects harmful resistance. Thus not only is harmful resistance fueled by weak governance, as shown in Table 3, but at the outset the market believes that such resistance reflects a lower probability the bid will be successful in imposing an opportunity cost on target shareholders. We repeat the analysis using the estimated probability of harmful resistance for model (1) in Table 3. A full interaction model akin to model (3) of Table 4 is estimated. We find the results (not tabulated) are comparable to those presented in Table 4.

How does the initial assessed market-based probability of success compare with actual success rates within the sample? Takeover attempts ultimately leading to a complete change in control for the target firm are identified as bids that are ultimately declared unconditional in all respects. This essentially means that the shareholders of the target company have agreed to sell to the bidder. The overall success rate for the total sample is roughly 70%, but this proportion is only 60% for the harmful resistance group compared to 77% for the remaining targets. If the market's initial beliefs about the probability of success are conditioned on information available at the time of the initial bid, and nothing changes to alter those beliefs (that is, expected managerial actions and the market environment prove to be true on average), then the ultimate success rate should roughly equal the initial expected success rate. The actual success rates for both groups suggests that other influences following the initial bid worked to increase the success rates for the two groups. We find that these factors are the attraction of a third party offer and whether the final offer is for cash, both of which have a positive and statistically significant influence on the probability of success. The average marginal effects on the probability of success (Z-statistics) for an estimated probit model of ultimate success (not tabulated) equal: third party offer, 0.368 (5.03); full cash offer, 0.305 (2.95). In contrast the average marginal effect on success of resistance choice that places the target in the harmful resistance group is not significantly different from zero. McFadden's R² for the

model equals 0.22. Thus while the market-wide initial belief is that resistance driven by harmful resistance reduces the probability of success, subsequent actions of third party bidders raise the probability of success. It is important however to remember that we are here referring to ultimate success rates and that 40% of those cases exhibiting harmful resistance ultimately fail.

6. C.E.O. turnover and reputation

Harmful resistance that destroys shareholder wealth is potentially an indicator of a more pervasive agency problem. Hence it can be a reflection of potentially unobservable variables or forces that characterize managerial self-interest or entrenchment. In this section we employ membership in the harmful resistance group as an indicator of a more deep-seated problem and explore whether the association between this indicator and events that arise subsequent to the takeover contests are consistent with the existence of such problems.

In Table 5 we use resistance choice as a potential indicator of a more deep-seated problem, which may include an agency problem but may also capture managerial quality or ability. Alternatively from the results presented in Table 3 we infer that the choice to adopt harmful resistance is associated with weak incentives and weak corporate governance. The extant literature suggests that the strength of corporate governance and managerial incentives influence the disciplining of managers of poorly performing firms through turnover (Weisbach, 1988; Kang and Shivdasani, 1995; Volpin, 2002). Therefore if membership in the harmful resistance group captures the presence of more deep-seated value-related problems within the firm and if internal and external forces move to correct the problem following a takeover contest, then we should see adjustments in the employment positions of the C.E.O.'s involved. We emphasize that we are using the classification variable here as an indicator or manifestation of managerial problems and as such we are not arguing that the resistance choice necessarily causes the changes in the employment opportunities of the C.E.O.'s in the sample. We examine three outcome variables as pertain to target company C.E.O.'s. The first is post-bid C.E.O. turnover. Here a binary

variable is used to identify cases where the C.E.O. either is turned over by the successful bidder (which can be a third party), or relinquishes office in the 12 months following a failed takeover attempt.

The general consensus in the literature is that weak performance is associated with C.E.O. turnover (Coughlan and Schmidt, 1985; Warner, Watts, and Wruck, 1988; Martin and McConnell, 1991; Dahya, McConnell and Travlos, 2002; Kini, Kracaw and Mian, 2004; Kaplan and Minton, 2012). We use the abnormal pre-bid run up in the target's share price over the 1 year period ending 30 days before (rumor of) the bid. Most studies of post-takeover target management turnover focus on successful offers. Agrawal and Walkling (1994) observe that target C.E.O.s are more likely to be replaced when a bid succeeds than when it fails. Our sample includes both successful and unsuccessful offers. We therefore control for whether the bidder is successful. The intuition being that if a takeover is completed (is successful) this provides the bidder with the opportunity to replace the target's management. We also include target size as a control variable as this is known to be correlated with numerous company characteristics.

In our sample we find that of the 54 cases in which C.E.O. turnover occurred 46 of those were associated with successful takeovers.

For all bids (completed or otherwise) 52% of the C.E.O.s represented in the harmful resistance group turn over compared to only 39% for the remaining targets. Even for C.E.O.s who employ harmful resistance but are still acquired there remains a 37% chance of being retained.¹⁹

The average marginal effects computed from a probit specification in which C.E.O. turnover is the dependent variable are presented in column (1) of Table 5. C.E.O. turnover is positively related to membership in the harmful resistance group consistent with such actions on the part of managers revealing self-interest as well as weak incentives and governance. This result suggests that turnover is associated with harmful resistance. Turnover is negatively related to prior performance and is positively

¹⁹ Bates and Becher (2012) only focus on failed bids. If we consider only the failed bids in our sample then the rate of C.E.O. turnover for the harmful resistance group (20 cases) is 35% compared to 6% for the remaining targets (16 cases). Furthermore, the market-adjusted returns from bid announcement through to 1 year post-bid (clean of any further bids) are -3.01% and 18.02% for the harmful resistance and remaining target groups, respectively.

related to bid success. The results on prior performance are consistent with those reported by others such as Martin and McConnell (1991) and Kini, Kracaw and Mian (2004) but suggest that managers that exhibit behaviors suggestive of shareholder neglect are more likely to be replaced. Harford (2003) also finds that a successful takeover attempt is one of the few important determinants in predicting the post-bid retention of inside directors. All average marginal effects aside from target size are statistically significant.

Moving from the position of a C.E.O. of a previously listed firm to a top manager of a subsidiary, or privately owned company, would likely not carry the same status or necessarily the same level of benefits, including control benefits. If the managerial labor market and the market for directors penalize C.E.O.s following harmful resistance then we should expect there to be a low likelihood that such C.E.O.s will secure top executive or director positions following a takeover (Fama and Jensen, 1983; Harford, 2003). We therefore divide the sample cases according to whether target C.E.O.s serve as the top executive of a listed firm in any of the 2 years following either bid completion or the 12 month period after the failure of a takeover attempt. Seventy-two percent (72%) of the target C.E.O.s in the harmful resistance group do not continue to run listed firms after resisting takeover bids. Interestingly the percentage is roughly the same for all other resisting targets in the sample.

In a similar fashion, we also gather information on whether the C.E.O.s of the target companies in our sample go on to serve on the boards per se of listed companies. We find that 58% do not hold a director position following a takeover contest. This percentage is also the same for the harmful resistance group and the other resisting targets. ²¹

²⁰ In which the eligible companies are members of a non-regulated industry (not in the financial/real estate, utility/telecommunication, public transport, broadcasting, and newspaper industries and have a full listing on the L.S.E..

²¹ If we consider only the failed bids in our sample then the rate of target C.E.O. attrition from the top executive positions of listed firms is 65% for the harmful resistance group compared to 38% for the remaining targets. The attrition rates for post-bid directorships are 55% and 19% for the harmful resistance and remaining target groups, respectively.

Columns (2) and (3) of Table 5 present results for models of the post-bid probability that a target C.E.O. will not (binary dependent variable takes the value 1) serve as a top executive or director, respectively, of a listed company. The explanatory variables are the same as those previously described for the model of C.E.O. turnover. The harmful resistance factor does not directly influence post-bid employment of a C.E.O. either as a top executive or as a director. However, if the bid is successful the impact is to increase the C.E.O.s probability of not holding such positions. Finally, C.E.O.s who are associated with larger targets face a decreased probability of not holding a position following a takeover. These results are statistically significant and are consistent with Harford (2003) who documents the potentially adverse impact of takeover bids on directors' wealth and future board seats.

Columns (4) and (5) of Table 5 drop the variable capturing whether the takeover bid is successful to instead investigate the relation between target C.E.O. turnover and their subsequent holding of a top executive position or a position as a director. These results indicate that the probability of not holding such positions following a takeover contest increases if the C.E.O. leaves the target company. The results are significant at approximately the 5% level for both cases. Agrawal and Walkling (1994) also conclude that target C.E.O.s who lose their jobs generally fail to find another senior executive position within a 1-3 year post-bid period.

7. Conclusions

The theoretical and empirical literature on takeovers is vast. Yet surprisingly little is known about the motivations and drivers behind the multiplicity of actions taken by managers to resist takeover bids. This paper attempts to address this deficiency by classifying resistance strategies according to whether they are harmful to stockholders. A distinction between resistance motivated by managerial self-interest/preservation and resistance motivated by shareholder wealth is fairly well grounded in the theoretical literature. Empirically, however, studies have either treated resistance as a homogenous response or else focused in isolation on predominantly pre-emptive, anti-takeover measures. In the spirit of Dann and DeAngelo (1988), we include all defensive asset restructuring and other blocking tactics in

our harmful resistance group provided these actions elicit a negative return stock price reaction on announcement. What we label as the harmful resistance group is then compared to the other resisting targets in our sample. The overall sample of 121 resisted bids is drawn from the 1989-2003 period in the U.K.. Conducting the study on U.K. resisted takeovers allows us to consistently capture managers' responses to bids in the absence of pre-emptive, anti-takeover mechanisms, made ineffectual under the U.K. rules.

Our analysis reveals new insights, in particular, into the role of incentive and governance structures of resisting firms. Firms in our sample resorting to harmful resistance strategies are found to have top managers with more pronounced ownership-based and age-related incentives for control, and other directors with equity interests less closely aligned to stockholders, as distinct from firms using resistance tactics designed to promote stockholder wealth. Firms that follow harmful resistance strategies have less independent boards, are exposed to weaker discipline from active outside blockholders, and are inferior performers. Prior studies have generally found no significant relation, or mixed results, when considering the effect of managerial incentives and corporate governance on either takeover resistance per se or the adoption of specific, anti-takeover measures. Managers are also found to pursue harmful resistance strategies in the presence of higher initial offer premiums and greater information availability about their firms. Previously it was known only that lower initial offer premiums were more likely to be resisted than recommended. Our observations on the nature of resistance are insensitive to an endogenous treatment of many of its determinants.

Furthermore, we find that the likelihood of resistance being motivated more by managerial self-interest is factored in adversely to announcement period returns as the capital market gauges its expectations of bids succeeding. The completion rate for resisted bids in our sample is ultimately influenced by the emergence and solicitation of third party offers, which are more prevalent among the other resisting firms in the sample. Finally, an incumbent C.E.O has a greater likelihood of being disciplined (turned over) when resisting for reasons that are harmful to shareholders. C.E.O. turnover has in turn an overriding detrimental effect on the aftermarket, board employment of these incumbent

managers. Previous studies have not documented an effect of takeover resistance in general on C.E.O. turnover that is incremental to pre-bid performance.

The continued relevance of the ongoing academic debate on takeover resistance is starkly illustrated by the controversy that persisted in the European Union until member states reached less than harmonious agreement on the extent to which anti-takeover mechanisms and bid resistance tactics should be controlled through the European Takeover Directive (2004/25/EC). Our study of the potential drivers and effects of resistance motivated by harmful resistance gives credence to those on both sides of the Atlantic who favour constraints on managerial actions driven by bad faith as distinct from those that legitimately enhance bargaining power on behalf of stockholders.

References

- Agrawal, A. and G.N. Mandelker, 1990. Large stockholders and the monitoring of managers: The case of anti-takeover charter amendments. *Journal of Financial and Quantitative Analysis* 25, 143-161.
- Agrawal, A. and R.A. Walkling, 1994. Executive careers and compensation surrounding takeover bids. *Journal of Finance* 49, 985-1014.
- Andrade, G., M. Mitchell and E. Stafford, 2001. New evidence and perspectives on mergers. *Journal of Economic Perspectives* 15, 103-120.
- Antia, M., C. Pantzalis and J. Park, 2010. CEO decision horizon and firm performance: An empirical investigation. *Journal of Corporate Finance*, 16, 288-301.
- Asquith, P., 1983. Merger bids, uncertainty, and stock returns. *Journal of Financial Economics* 11, 51-83.
- Bhagat, S. and B. Black, 1999. The uncertain relationship between board composition and firm performance. *The Business Lawyer* 54, 921–63.
- Bhagat, S.and B. Black, 2002. The non-correlation between board independence and long term firm performance. *Journal of Corporation Law* 27, 231–274.
- Bange, M.M. and M.A. Mazzeo, 2004. Board composition, board effectiveness and the observed form of takeover bids. *Review of Financial Studies* 17, 1185-1215.
- Bates, T.W. and D.A. Becher, 2012. Resistance by Takeover Targets: Managerial Bargaining or Bad Faith? Working Paper, Arizona State University.
- Bebchuk, L.A., 2002. The case against board veto in corporate takeovers. *University of Chicago Law Review* 69, 973-1035.
- Bebchuk, L.A., J.C. Coates IV and G. Subramanian, 2002. The powerful antitakeover force of staggered boards: Theory, evidence, and policy. *Stanford Law Review* 54, 887-952.
- Bebchuk, L.A. and A. Cohen, 2005. The costs of entrenched boards. *Journal of Financial Economics* 78, 409-433.
- Becht, M., P. Bolton and A. Roell, 2005. Corporate governance and control. *ECGI Working Paper Series in Finance*.
- Berkovitch, E., M. Bradley and N. Khanna, 1989. Tender offer auctions, resistance strategies, and social welfare. *Journal of Law, Economics and Organization* 5, 395-412.
- Berkovitch, E. and N. Khanna, 1990. How target shareholders benefit from value-reducing defensive strategies in takeovers. *Journal of Finance* 55, 137-156.
- Betton, S.and B.E. Eckbo, 2000. Toeholds, bid jumps, and expected payoffs in takeovers. *Review of Financial Studies* 13, 841–882.
- Betton, S., B.E. Eckbo and K.S. Thorburn, 2009. Merger negotiations and the toehold puzzle. *Journal of Financial Economics* 91, 158-178.
- Betton, S., B.E. Eckbo and K.S. Thorburn, 2010. Corporate takeovers, in, B.E. Eckbo (ed.), *Handbook of Corporate Finance: Empirical Corporate Finance*, Volume 2, Elsevier.
- Boone, A.L. and J.H. Mulherin, 2007. How are firms sold? Journal of Finance 62, 847-875.
- Bradley, M., A. Desai and E.H. Kim, 1983. The rationale behind interfirm tender offers: Information or synergy? *Journal of Financial Economics* 11, 183-206.

- Brickley, J.A., J.L. Coles and R.L. Terry, 1994. Outside directors and the adoption of poison pills. *Journal of Financial Economics* 35, 371-390.
- Brickley, J.A., R.C. Lease and C.W. Smith, 1988. Ownership structure and voting on anti-takeover charter amendments. *Journal of Financial Economics* 20, 267-291.
- Coles, J.L., N.D. Daniel and L. Naveen, 2008. Boards: Does one size fit all? *Journal of Financial Economics* 87, 329-356.
- Comment, R. and G.W. Schwert, 1995. Poison or placebo? Evidence on the deterrence and wealth effects of modern anti-takeover measures. *Journal of Financial Economics* 39, 3-43.
- Cotter, J.F., A. Shivdasani and M. Zenner, 1997. Do independent directors enhance target stockholder wealth during tender offers? *Journal of Financial Economics* 43, 195-218.
- Cotter, J.F. and M. Zenner, 1994. How managerial wealth affects the tender offer process. *Journal of Financial Economics* 35, 63-97.
- Coughlan, A.T. and R.M. Schmidt, 1988. Executive compensation, managerial turnover and firm performance: An empirical investigation. *Journal of Accounting and Economics* 7, 43-66.
- Dahya, J., J.J. McConnell and N.G. Travlos, 2002. The Cadbury Committee, corporate performance and top management turnover. *Journal of Finance* 57, 461-483.
- Dann, L.Y. and H. DeAngelo, 1988. Corporate financial policy and corporate control: A study of defensive adjustments in asset and ownership structure. *Journal of Financial Economics* 20, 87-127
- DeAngelo, H., and E.M. Rice, 1983. Antitakeover charter amendments and stockholder wealth. *Journal of Financial Economics* 11, 329-359.
- Dechow, P. and R. Sloan, 1991. Executive incentives and the horizon problem: An empirical investigation. *Journal of Accounting and Economics* 14, 51–89.
- Demsetz, H., 1983. The structure of ownership and the theory of the firm. *Journal of Law and Economics* 26, 375-90.
- Demsetz, H. and K. Lehn, 1985. The structure of corporate ownership: Causes and consequences. *Journal of Political Economy* 93, 1155-1177.
- Demsetz, H. and B. Villalonga, 2001. Ownership structure and corporate performance. *Journal of Corporate Finance* 7, 209-33.
- Denis, D.J., 1990. Defensive changes in corporate payout policy: Share repurchases and special dividends. *Journal of Finance* 45, 1433-1456.
- Fabozzi, F.J., M.G. Ferri, T. Fabozzi and J. Tucker, 1988. A note on unsuccessful tender offers and stockholder returns. Journal of Finance 63, 1275-1283.
- Faccio, M. and R.Masulis, 2005. The choice of payment method in European mergers and acquisitions. *Journal of Finance* 60, 1345-1388.
- Faleye, O., 2007. Classified boards, firm value, and managerial entrenchment. Journal of Financial Economics 83, 501-529.
- Fama, E. and M. Jensen, 1983. Separation of ownership and control. *Journal of Law and Economics* 26, 301-325.
- Fishman, M.J., 1988. A theory of preemptive takeover bidding. Rand Journal of Economics 19, 88-101.

- Fishman, M.J., 1989. Preemptive bidding and the role of the medium of exchange in acquisitions. *Journal of Finance* 44, 41-57.
- Forrester, C. and C. Ferber, 2012. *Fiduciary Duties and Other Responsibilities of Corporate Directors and Officers*. Morrison & Foerster LLP and R. R. Donnelley, New York.
- Franks, J. and R. Harris, 1989. Shareholder wealth effects of corporate takeovers: The U.K. experience 1955-1985. *Journal of Financial Economics* 23, 225-245.
- Franks, J. and C. Mayer, 1996. Hostile takeovers and the correction of managerial failure. *Journal of Financial Economics* 40, 163-181.
- Giammarino, R.M. and R.L. Heinkel, 1986. A model of dynamic takeover behaviour. *Journal of Finance* 41, 465-480.
- Gibbons, R. and K. Murphy, 1992. Optimal incentive contracts in the presence of career concerns: Theory and evidence. *Journal of Political Economy* 100, 468–505.
- Gillan, S., 2006. Recent developments in corporate governance: An overview. *Journal of Corporate Finance* 12, 381-402.
- Goldman, E. and J. Qian, 2005. Optimal toeholds in takeover contests. *Journal of Financial Economics* 77, 321-346.
- Gompers, P., J. Ishii and A. Metrick, 2003. Corporate governance and equity prices. *Quarterly Journal of Economics* 118, 107–155.
- Harford, J., 2003. Takeover bids and target directors' incentives: The impact of a bid on directors' wealth and board seats. *Journal of Financial Economics* 69, 51-83.
- Harford, J., M. Humphery-Jenner and R. Powell, 2012. The sources of value destruction in acquisitions by entrenched managers. *Journal of Financial Economics* 106, 247-261.
- Hermalin, B.E. and M.S. Weisbach, 2003. Boards of directors as an endogenously determined institution: A survey of the economic literature. *Economic Policy Review* April, 7-26.
- Heron, R.A. and E. Lie, 2006. On the use of poison pills and defensive payouts by takeover targets. *Journal of Business* 79, 1783-1807.
- Hirshleifer, D. and I.P.L. Png, 1989. Facilitation of competing bids and the price of a takeover target. *Review of Financial Studies* 2, 587-606.
- Hirshleifer, D. and A.V. Thakor, 1994. Managerial performance, boards of directors and takeover bidding. *Journal of Corporate Finance* 1, 63-90.
- Holderness, C., 2009. The myth of diffuse ownership in the United States. *Review of Financial Studies* 22, 1377-1408.
- Holmström, B., 1982. Managerial incentive schemes: A dynamic perspective, in *Essays in Economics and Management in Honour of Lars Wahlbeck*, Swenska Handelshögkolan, Helsinki.
- Huang, Y.-S. and R.A. Walkling, 1987. Target abnormal returns associated with acquisition announcements: Payment, acquisition form and managerial resistance. *Journal of Financial Economics* 19, 329-349.
- Jarrell, G.A., J.A. Brickley and J.M. Netter, 1988. The market for corporate control: The empirical evidence since 1980. *Journal of Economic Perspectives* 2, 49-68.
- Jarrell, G.A. and A. Poulsen. 1987. Shark repellents and stock prices: The effects of antitakeover amendments since 1980. *Journal of Financial Economics* 19, 127-168.

- Jennings, R.H. and M.A. Mazzeo, 1993. Competing bids, target management resistance and the structure of takeover bids. *Review of Financial Studies* 6, 883-909.
- Jensen, M. C., 1986. Agency costs of free cash flow, corporate finance and takeovers. *American Economic Review* 76, 323–329.
- Jensen, M.C., 1993. The modern industrial revolution, exit and the failure of internal control systems. *Journal of Finance* 48, 831-880.
- Jensen, M.C. and R.S. Ruback, 1983. The market for corporate control: The scientific evidence. *Journal of Financial Economics* 11, 5-50.
- Jenter, D. and K. Lewellen, 2011. C.E.O. preferences and acquisitions. Working paper, Stanford University.
- Kang, J. and A. Shivdasani, 1995. Firm performance, corporate governance, and top executive turnover in Japan. *Journal of Financial Economics* 38, 29-58.
- Kaplan, S.N. and B.A. Minton, 2012. How has C.E.O. turnover changed? *International Review of Finance* 12, 57-87.
- Kaplan, S.N. and D. Reishus, 1990. Outside directorships and corporate performance. *Journal of Financial Economics* 27, 389-410.
- Kini, O., W. Kracaw and S. Mian, 2004. The nature of discipline by corporate takeovers. *Journal of Finance* 59, 1511-1552.
- Linn, S.C. and J.J. McConnell, 1983. An empirical investigation of the impact of `antitakeover' amendments on common stock prices. *Journal of Financial Economics* 11,361-99.
- Long, J. S. and J. Freese, 2006. *Regression Models for Categorical Dependent Variables Using Stata*, Stata Press, College Station, Texas.
- Manne, H.G., 1965. Mergers and the market for corporate control. *Journal of Political Economy* 73, 110-120.
- Martin, K. and J.J. McConnell, 1991. Corporate performance, corporate takeovers and management turnover. *Journal of Finance* 46, 671-687.
- Martin, K., 1996. The method of payment in corporate acquisitions, investment opportunities, and management ownership. *Journal of Finance* 51, 1227-1246.
- Masulis, R.W., C. Wang and F. Xie, 2007. Corporate governance and acquirer returns. *Journal of Finance* 62, 1851–1889.
- Mikkelson, W.H. and M.M. Partch, 1989. Managers' voting rights and corporate control. *Journal of Financial Economics* 25, 263-290.
- Moeller, T., 2005. Let's make a deal! How stockholder control impacts merger payoffs. *Journal of Financial Economics* 76, 167-190.
- Morck, R., A. Shleifer and R.W. Vishny, 1988. Characteristics of targets of hostile and friendly takeovers, in A.J. Auerbach (ed.), *Corporate Takeovers: Causes and Consequences*, University of Chicago Press, Chicago.
- Newey, W. K., 1987. Efficient estimation of limited dependent variable models with endogenous explanatory variables. *Journal of Econometrics* 36, 231-250.
- Ruback, R.S., 1988. Do target stockholders lose in unsuccessful control contests?, in A.J. Auerbach, (ed.), *Corporate Takeovers: Causes and Consequences*, University of Chicago Press, Chicago.

- Schwert, G.W., 2000. Hostility in takeovers: In the eyes of the beholder? *Journal of Finance* 55, 2599-2640.
- Serfling, M.A., 2013. C.E.O. age and the riskiness of corporate policies. Working Paper, University of Arizona.
- Shivdasani, A., 1993. Board composition, ownership structure and hostile takeovers. *Journal of Accounting and Economics* 16, 167-198.
- Shleifer, A. and R.W. Vishny, 1986. Large stockholders and corporate control. *Journal of Political Economy* 94, 461-488.
- St-Pierre, J., J.-M. Gagnon and J. Saint-Pierre, 1996. Concentration of voting rights and board resistance to takeover bids. *Journal of Corporate Finance* 3, 45-73.
- Stout, L., 2002. Do antitakover defenses reduce shareholder wealth? The ex-ante/ex-post measurement problem. *Stanford Law Review* 55, 845-861.
- Stulz, R.M., 1988. Managerial control of voting rights: Financing policies and the market for corporate control. *Journal of Financial Economics* 20, 25-54.
- Volpin, P.F., 2002. Governance with poor investor protection: Evidence from top executive turnover in Italy. *Journal of Financial Economics* 64, 61-90.
- Walkling, R.A., 1985. Predicting tender offer success: A logistic analysis. *Journal of Financial and Quantitative Analysis* 20, 461-478.
- Walkling, R.A. and M.S. Long, 1984. Agency theory, managerial welfare and takeover bid resistance. *Rand Journal of Economics* 15, 54-68.
- Warner, J.B., R.L. Watts and K.H. Wruck, 1988. Stock prices and top management changes. *Journal of Financial Economics* 20, 461-492.
- Weisbach, M.S., 1988. Outside directors and C.E.O. turnover. *Journal of Financial Economics* 20, 431-460.
- Wooldridge, J., 2002. *Econometric Analysis of Cross Section and Panel Data*, Cambridge, Massachusetts: MIT Press.
- Yermack, D., 1996. Higher market valuation of companies with a smaller board of directors. *Journal of Financial Economics* 40, 185-211.

Table 1
Resistance tactics.

Percentage of targets using specific resistance tactics for the subsample of 50 resisted U.K. takeover bids classified as members of the harmful resistance group from a sample of 121 publicly resisted U.K. takeover bids from the period 1989-2003. Membership in the harmful resistance group of resisting targets is defined by a target experiencing a negative share price reaction at the announcement of the resistance action for each of the tactics listed. Resistance tactics: (1) Spin-offs/sell-offs which are divestments that deny bidders access to assets of value; (2) Mergers/acquisitions/joint ventures which make target firms more cumbersome to acquire due to size, strategic or antitrust grounds, or which eliminate bidders directly; (3) Stock repurchases/special dividends which involve payouts that effectively eliminate bidders' plans for the utilization of excess cash held by the target, and, in the case of targeted repurchases, that increase the proportion of stock in the hands of friendly stockholders; (4) Going private transactions which use competing management buyouts to create potentially costly bidding contests; (5) Management changes which make removal of newly appointed officers especially costly because of special contractual payments ('golden parachutes') triggered by a takeover; and (6) White squires which involve solicitation of friendly third party participants who acquire a stake that will effectively block the takeover. Market-adjusted returns at announcement of the resistance actions use the F.T.S.E All Share index as the market proxy.

	1989-2003 (50 bids)	Average market- adjusted return at announcement of action (%)	% of market- adjusted returns that are <0	1989-1996 (24 bids)	1997-2003 (26 bids)
Resistance tactic	(1)	(2)	(3)	(4)	(5)
Spin-off/sell-off	46.0%	-2.72	100	58.3%	34.6%
Merger/acquisition/joint venture	30.0%	-3.46	100	25.0%	34.6%
Stock repurchase/special dividend	16.0%	-1.24	100	8.3%	23.1%
Going-private transaction	10.0%	-1.36	100	4.2%	15.4%
Management change	10.0%	-0.77	100	12.5%	7.7%
White squire	8.0%	-1.37	100	8.3%	7.7%

Table 2

Descriptive statistics.

Descriptive statistics for target firm and initial offer characteristics by resistance strategy for a sample of 121 publicly resisted U.K. takeover bids 1989-2003. Membership in the harmful resistance group is as described in Table 1. Target/initial offer characteristics; C.E.O. ownership (%) is the percentage of outstanding equity at the bid announcement date in which the most influential target officer (based on title, and, if no officer carries the title 'Chief Executive Officer (C.E.O.)', on compensation) has a beneficial interest. C.E.O. age is the age of the target C.E.O. at bid announcement. Other directors' ownership (%) is the percentage of outstanding equity at the bid announcement date in which target directors other than the C.E.O. have a (non-duplicated) beneficial interest. Independent chairperson (=1) is a binary variable equal to one if the director serving as target chairperson at the bid announcement date is non-aligned (no personal, employment, or business connections to affect their oversight) and holds at least one other directorship, that is noninterlocking (we permit the non-aligned chairperson to serve as an executive director elsewhere provided the target C.E.O. is not on the same board), in a fully listed firm outside the financial/real estate, utility/telecommunication, public transport, broadcasting, and newspaper industries. Independent directors (%) is the proportion of independent (as described for Independent chairperson) target board members (including the C.E.O.) at the bid announcement date. Board size is the number of target firm directors (including the C.E.O.) at the bid announcement date. Outside blockholdings (%) is the percentage sum of non-aligned (as described for Independent chairperson) stock interests wielding at least 5% of total target equity votes (excluding initial bidder toeholds) at the bid announcement date. Offer premium is the percentage difference between the unit initial offer value (using the bidder's stock price 5 trading days before the bid announcement date for offers involving an exchange of common stock) and the target firm's stock price (adjusted for rumor as per the Valuation ratio described below). Bidder toehold (%) is the percentage of total target equity votes controlled by the initial bidder and any connected parties immediately prior to offer announcement. Bidder/dissident director (=1) is a binary variable equal to one if the initial bidder has a representative on the target board, or a dissident target director pledges support for this bidder. Full cash terms (=1) is a binary variable equal to one if the initial bid offers full cash terms (including a full cash alternative to the main terms). Bidder approach (=1) is a binary variable equal to one if the initial bidder makes a genuine approach to the target board before launching their bid. Valuation ratio is the average market to tangible book value of assets (where market value of assets is the common stock price multiplied by the number of outstanding shares plus the book value of non-equity securities and total debt, and the denominator is total assets minus intangibles) of the target firm for the two financial years preceding the bid announcement date (adjusting the more recent market value of equity to just before an unofficial rumor period of 30 trading days plus any official rumor period). Asset turnover is the average sales to tangible assets ratio of the target firm for the two financial years preceding the bid announcement date. Young firm (=1) is a binary variable equal to one if the target firm has been fully listed for 6 years or less at the bid announcement date. Sample size is determined by data availability. **, * indicates significance at the 1% and 5% levels, respectively. (Table continued on following page).

40

Table 2 (continued)

	Targets	s in the har gro	rmful resistance up	All	other resis	ting targets		Difference		
Target/initial offer characteristics	Mean (1)	Media (2)	Proportion of targets/offers (3)	Mean (4)	Median (5)	Proportion of targets/offers (6)	t-stat. for mean diff. (7)	Kruskal-Wallis χ^2 med. diff. (8)	Z-stat. for prop. diff. (9)	Sample size (10)
C.E.O. ownership (%)	2.30	0.09		1.47	0.14		0.83	-0.03		121
C.E.O. age	52.02	53.00		51.24	51.00		0.86	-0.78		120
Other directors' ownership (%)	1.24	0.30		2.89	0.44		-1.62	-0.67		121
Independent chairperson (=1)			0.20			0.45			-2.85**	121
Independent directors (%)	16.61	16.67		18.74	16.67		-0.78	-0.71		121
Board size	7.18	7.00		7.10	7.00		0.19	0.00		121
Outside blockholdings (%)	25.05	24.50		32.23	33.00		-1.99*	-3.97*		121
Offer premium (%)	39.34	31.47		30.89	28.04		1.07	-0.36		120
Bidder toehold (%)	11.52	1.00		15.23	5.00		-1.19	-1.09		121
Bidder/dissident director (=1)			0.06			0.17			-1.79	121
Full cash terms (=1)			0.62			0.75			-1.49	121
Bidder approach (=1)			0.34			0.25			1.03	121
Valuation ratio	1.27	1.17		1.33	1.18		-0.69	-0.12		115
Asset turnover	1.29	1.20		1.46	1.43		-0.96	-1.71		115
Young firm (=1)			0.06			0.24			-2.62**	121

Table 3

Resistance choice.

Average marginal effects of explanatory variables on the probability of target board resistance reflecting harmful resistance for a sample of 121 publicly resisted U.K. takeover bids 1989-2003. Absolute values of z-statistics are presented (in parentheses) for tests of the null hypothesis that the average marginal effect is equal to zero and are computed using robust standard errors. Base model from which the marginal effects are computed is a probit specification in which the dependent variable takes the value 1 if the target belongs to the harmful resistance group and 0 otherwise. Columns labeled Probit treat Offer premium as exogenous by assumption. Columns labeled 2-Stage treat Offer premium as endogenous and present Wald tests statistics of the null hypothesis that Offer premium is exogenous. Pre-bid run-up of the target's stock (described below) and the natural logarithm of target firm size (market value of assets of the target firm for the financial year preceding the bid announcement date, where market value of assets is the common stock price, adjusted to just before an unofficial rumor period of 30 trading days plus any official rumor period, multiplied by the number of outstanding shares plus the book value of non-equity securities and total debt). P-values > 0.05 indicate the null of exogeneity is not rejected at that level. 2-stage model estimated by maximum likelihood methods. Wald test of model statistics test the null hypothesis that the estimated coefficients of the model are jointly equal to zero. Refer to Table 1 for definitions of target board resistance reflecting harmful resistance and to Table 2 for definitions of variables. Pre-bid run-up (%) is the daily cumulative market (F.T.S.E. All Share index) adjusted target stock return over the 12 month period ending 30 trading days either prior to the earliest release concerning a bid rumor, or the initial offer announcement if there is no formal disclosure anticipating a takeover attempt. 1989-1996 bid (=1) takes the value 1 if the initial bid is publicly resisted during the period 1989-1996 and 0 otherwise. Sample size is determined by data availability. **, * indicates significance at the 1% and 5% levels, respectively.

(Table continued on following page).

Table 3 (continued)

	Dependent variable: Targets in the harmful resistance group = 1 , otherwise = 0							
	Probit	2-Stage	Probit	2-Stage	Probit	2-Stage	Probit	2-Stage
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
C.E.O. ownership (%)	0.044	0.035	0.042	0.030	0.041	0.030	0.036	0.027
2	(4.66)**	(2.88)**	(4.13)**	(2.89)**	(4.12)**	(2.81)**	(4.07)**	(2.92)**
C.E.O. age	0.239	0.226	0.212	0.206	0.225	0.216	0.227	0.219
-	(2.63)**	(2.51)*	(2.29)*	(2.28)*	(2.42)*	(2.39)*	(2.53)*	(2.51)*
$(C.E.O. age)^2$	-0.022	-0.020	-0.019	-0.019	-0.021	-0.020	-0.021	-0.020
	(2.47)*	(2.34)*	(2.17)*	(2.14)*	(2.29)*	(2.24)*	(2.41)*	(2.36)*
Other directors' own (%)	-0.078	-0.082	-0.076	-0.083	-0.076	-0.081	-0.078	-0.084
` '	(4.39)**	(4.83)**	(3.96)**	(4.56)**	(4.26)**	(4.72)**	(4.39)**	(4.69)**
Ind. chair (=1)	-0.240	-0.248	-0.204	-0.217	` ,	` ,	` ,	` ,
` ,	(2.84)**	(2.92)**	(2.68)**	(2.77)**				
Board size	-0.025	-0.023	-0.026	-0.026	-0.025	-0.024	-0.023	-0.023
	(1.25)	(1.17)	(1.43)	(1.39)	(1.28)	(1.21)	(1.33)	(1.32)
Outside blockholdings (%)	-0.007	-0.007	-0.008	-0.008	-0.008	-0.008	-0.007	-0.006
8 ()	(2.92)**	(2.97)**	(3.41)**	(3.63)**	(3.84)**	(3.87)**	(2.95)**	(2.94)**
Valuation ratio	-0.141	-0.164	-0.170	-0.208	-0.172	-0.206	-0.160	-0.192
	(1.83)	(2.12)*	(2.11)*	(2.56)*	(2.02)*	(2.42)*	(2.13)*	(2.42)*
Asset turnover	-0.165	-0.155	-0.170	-0.162	-0.170	-0.161	-0.181	-0.171
	(3.37)**	(3.17)**	(3.24)**	(3.36)**	(3.55)**	(3.50)**	(3.84)**	(3.79)**
Young firm (=1)	-0.546	-0.510	-0.509	-0.468	-0.515	-0.479	-0.507	-0.473
	(5.04)**	(3.95)**	(5.46)**	(4.48)**	(5.34)**	(4.50)**	(5.81)**	(5.15)**
Offer premium (%)	0.004	0.003	0.003	0.003	0.003	0.003	0.003	0.003
1	(3.41)**	(2.31)*	(3.09)**	(2.27)*	(3.10)**	(2.29)*	(2.64)**	(2.09)*
Bidder toehold (%)	-0.004	-0.004	(2007)	(=)	(===)	(====)	(====)	(=101)
	(1.54)	(1.36)						
Full cash terms (=1)	-0.193	-0.194	-0.159	-0.162	-0.157	-0.158	-0.131	-0.125
	(2.20)*	(2.22)*	(1.87)	(1.88)	(1.81)	(1.80)	(1.65)	(1.56)
Bidder approach (=1)	0.095	0.107	0.076	0.095	0.093	0.109	0.099	0.112
	(1.00)	(1.11)	(0.87)	(1.07)	(1.05)	(1.23)	(1.17)	(1.32)
Pre-bid run-up (%)	0.001	0.001	0.001	0.001	0.002	0.001	0.001	0.001
	(1.20)	(0.91)	(1.34)	(1.17)	(1.66)	(1.45)	(1.42)	(1.35)

Table 3 (continued)

	Dependent variable: Targets in the harmful resistance group $= 1$, otherwise $= 0$							
	Probit (1)	2-Stage (2)	Probit (3)	2-Stage (4)	Probit (5)	2-Stage (6)	Probit (7)	2-Stage (8)
1989-1996 bid. (=1)	-0.304 (3.66)**	-0.304 (3.77)**	-0.283 (3.19)**	-0.290 (3.49)**	-0.307 (3.54)**	-0.308 (3.65)**		
Bidder/dissident dir. (=1)	(3133)	(0111)	-0.224	-0.240	-0.238	-0.256	-0.285	-0.302
Independent directors (%)			(1.94)	(1.97)*	(1.98)* -0.005 (1.60)	(2.03)* -0.005 (1.51)	(2.55)*	(2.63)**
Ind. dir (%) x 1989-1996					(====)	(=== = /	-0.012	-0.012
bid (=1)							(2.99)**	(3.07)**
McFadden's R ²	0.38		0.39		0.37		0.37	(3.07)
Wald test of model	39.58	41.02	37.22	40.25	41.72	40.13	41.18	38.62
p-value	0.00	0.00	0.01	0.00	0.00	0.00	0.00	0.01
Observations	114	113	114	113	114	113	114	113
Wald test for exogeneity of Offer premium		1.26		1.36		1.16		0.40
p-value		0.26		0.24		0.28		0.53

Table 4
Market assessment of the probability of bid success.

Linear regressions of announcement period cumulative abnormal returns (C.A.R.s) for a sample of 121 publicly resisted U.K. takeover bids from the period 1989-2003. Announcement period C.A.R.s (in %) are computed as daily cumulative abnormal (F.T.S.E. All Share index adjusted) target stock returns over the period from 30 trading days before bid rumor, or formal announcement if no prior 12 month rumor related disclosure, through to the initial offer announcement date. Harmful resistance is a binary variable equal to one if the target is classified as described in Table 1 and 0 otherwise. Target size (ln) is as defined in Table 3. Offer premium (%), bidder toehold (%) and Full cash terms (=1) are as defined in Table 2. Sample size is determined by data availability. T-statistics are reported in parentheses and are computed using robust standard errors. **, * indicates significance at the 1% and 5% levels, respectively. Assessed probability the offer will be successful is shown in bold

Dependent variable:
Target company announcement period CARs

	Harmful	All others	Full interaction
	resistance group		model
	(1)	(2)	(3)
Offer premium (%)	0.422	0.654	0.654
	(9.83)**	(13.23)**	(12.48)**
Bidder toehold (%)	-0.394	-0.234	-0.234
Bidder toehold (%)	(2.88)**	(2.75)**	(2.59)*
Full cash terms (=1)	16.464	3.973	3.973
	(3.97)**	(1.14)	(1.08)
Target size (ln)	3.438	1.145	1.145
	(2.91)**	(1.17)	(1.11)
Harmful resistance (=1)			-47.263
			(1.53)
Harmful resistance (=1) x Off. Premium (%)			-0.232
			(3.52)**
Harmful resistance (=1) x Bidder toehold (%)			-0.161
			(1.03)
Harmful resistance (=1) x Full cash terms (=1)			12.491
			(2.35)*
Harmful resistance (=1) x Target size (ln)			2.293
· / · · · · · · · · · · · · · · · · · ·			(1.52)
Constant	-65.925	-18.662	-18.662
	(2.69)**	(0.94)	(0.89)
Adj. R-squared	0.74	0.76	0.75
F-statistic	34.73	54.36	39.49
p-value	0.00	0.00	0.00
Observations	49	70	119

Table 5 Post-bid C.E.O. outcomes.

Average marginal effects of explanatory variables on the probability of (1) Chief Executive Officer (C.E.O.) turnover, (2) C.E.O. non-retention of a position as a top executive at the target or another firm, and (3) C.E.O. non-retention of a position as a director at the target firm or another firm. C.E.O. turnover is a binary variable that takes the value 1 if the target C.E.O. either is turned over by the successful bidder, or relinquishes office in the 12 months following a failed takeover attempt, and 0 otherwise. C.E.O. non-retention of a top executive position is a binary variable that takes the value 1 if the target C.E.O. does not go on to be a top executive of a fully listed firm (outside the financial/real estate, utility/telecommunication, public transport, broadcasting, and newspaper industries) in any of the 2 years following either bid completion or the 12 month period after the failure of a takeover attempt, and 0 otherwise. C.E.O. nonretention of a director position is a binary variable that takes the value 1 if the target C.E.O. does not go on to be a director of a fully listed firm (from a non-regulated industry as defined above) in any of the 2 years following either bid completion or the 12 month period after the failure of a takeover attempt, and 0 otherwise. The sample is composed of 121 publicly resisted U.K. takeover bids from the period 1989-2003. Absolute values of z-statistics are presented (in parentheses) for tests of the null hypothesis that the average marginal effect is equal to zero and are computed using robust standard errors. Base model from which marginal effects are computed is a probit specification in which the dependent variable takes the value 1 or 0 as described above. Harmful resistance (=1) is as defined in Table 1. Pre-bid run-up (%) and Target size (ln) are as defined in Table 3. Bid successful (=1) is a binary variable equal to one if the prevailing bid (that can be from a third party) is ultimately declared unconditional in all respects. Sample size is determined by data availability. **, * indicates significance at the 1% and 5% levels, respectively.

	C.E.O. turnover	C.E.O. without post-bid top executive position	C.E.O. without post-bid directorship	C.E.O. without post-bid top executive position	C.E.O. without post-bid directorship
-	(1)	(2)	(3)	(4)	(5)
Harmful resistance (=1)	0.204 (2.22)*	0.085 (0.95)	0.127 (1.42)	0.016 (0.21)	0.051 (0.61)
Pre-bid run-up (%)	-0.003 (2.55)*	-0.001 (1.03)	-0.001 (0.82)	-0.001 (1.02)	-0.001 (0.81)
Bid successful (=1)	0.339 (3.74)**	0.275 (3.01)**	0.272 (3.11)**		
Target size (ln)	0.021 (0.74)	-0.063 (2.80)**	-0.114 (5.71)**	-0.061 (2.94)**	-0.109 (5.41)**
C.E.O. turnover (=1)				0.160 (2.03)*	0.161 (1.95)
McFadden's R ² Wald test p-value	0.12 20.64 0.00	0.13 16.07 0.00	0.18 26.82 0.00	0.09 13.69 0.01	0.15 22.29 0.00
Observations	120	120	120	120	120

crm working paper series



CFR Working Papers are available for download from www.cfr-cologne.de.

Hardcopies can be ordered from: Centre for Financial Research (CFR), Albertus Magnus Platz, 50923 Koeln, Germany.

2014

No.	Author(s)	Title
14-01	N. F. Carline, S. C. Linn, P. K. Yadav	Corporate Governance and the Nature of Takeover Resistance
2013		
No.	Author(s)	Title
13-11	R. Baule, O. Korn, S. Saßning	Which Beta is Best? On the Information Content of Option-implied Betas
13-10	V. Agarwal, L. Ma	Managerial Multitasking in the Mutual Fund Industry
13-09	M. J. Kamstra, L.A. Kramer, M.D. Levi, R. Wermers	Seasonal Asset Allocation: Evidence from Mutual Fund Flows
13-08	F. Brinkmann, A. Kempf, O. Korn	Forward-Looking Measures of Higher-Order Dependencies with an Application to Portfolio Selection
13-07	G. Cici, S. Gibson, Y. Gunduz, J.J. Merrick, Jr.	Market Transparency and the Marking Precision of Bond Mutual Fund Managers
13-06	S.Bethke, A. Kempf, M. Trapp	The Correlation Puzzle: The Interaction of Bond and Risk Correlation
13-05	P. Schuster, M. Trapp, M. Uhrig-Homburg	A Heterogeneous Agents Equilibrium Model for the Term Structure of Bond Market Liquidity
13-04	V. Agarwal, K. Mullally, Y. Tang, B. Yang	Mandatory Portfolio Disclosure, Stock Liquidity, and Mutual Fund Performance
13-03	V. Agarwal, V. Nanda, S.Ray	Institutional Investment and Intermediation in the Hedge Fund Industry
13-02	C. Andres, A. Betzer, M. Doumet, E. Theissen	Open Market Share Repurchases in Germany: A Conditional Event Study Approach
13-01	J. Gaul, E. Theissen	A Partially Linear Approach to Modelling the Dynamics of Spot and Futures Prices

No.	Author(s)	Title
12-12	Y. Gündüz, J. Nasev, M. Trapp	The Price Impact of CDS Trading
12-11	Y. Wu, R. Wermers, J. Zechner	Governance and Shareholder Value in Delegated Portfolio Management: The Case of Closed-End Funds
12-10	M. Trapp, C. Wewel	Transatlantic Systemic Risk
12-09	G. Cici, A. Kempf, C. Sorhage	Are Financial Advisors Useful? Evidence from Tax-Motivated Mutual Fund Flows
12-08	S. Jank	Changes in the composition of publicly traded firms: Implications for the dividend-price ratio and return predictability
12-07	G. Cici, C. Rosenfeld	The Investment Abilities of Mutual Fund Buy-Side Analysts
12-06	A. Kempf, A. Pütz, F. Sonnenburg	Fund Manager Duality: Impact on Performance and Investment Behavior
12-05	R. Wermers	Runs on Money Market Mutual Funds
12-04	R. Wermers	A matter of style: The causes and consequences of style drift in institutional portfolios
12-02	C. Andres, E. Fernau, E. Theissen	Should I Stay or Should I Go? Former CEOs as Monitors
12-01	L. Andreu, A. Pütz	Are Two Business Degrees Better Than One? Evidence from Mutual Fund Managers' Education
		Evidence nom widthan und managers Education
2011		Evidence nom Muldam und Managers Education
2011 No.	Author(s)	Title
	Author(s) V. Agarwal, JP. Gómez, R. Priestley	
No.	V. Agarwal, JP. Gómez,	Title Management Compensation and Market Timing under Portfolio
No. 11-16	V. Agarwal, JP. Gómez, R. Priestley	Title Management Compensation and Market Timing under Portfolio Constraints Can Internet Search Queries Help to Predict Stock Market
No. 11-16 11-15	V. Agarwal, JP. Gómez, R. Priestley T. Dimpfl, S. Jank P. Gomber, U. Schweickert,	Title Management Compensation and Market Timing under Portfolio Constraints Can Internet Search Queries Help to Predict Stock Market Volatility? Liquidity Dynamics in an Electronic Open Limit Order Book:
No. 11-16 11-15 11-14	V. Agarwal, JP. Gómez, R. Priestley T. Dimpfl, S. Jank P. Gomber, U. Schweickert, E. Theissen	Title Management Compensation and Market Timing under Portfolio Constraints Can Internet Search Queries Help to Predict Stock Market Volatility? Liquidity Dynamics in an Electronic Open Limit Order Book: An Event Study Approach Irrationality or Efficiency of Macroeconomic Survey Forecasts?
No. 11-16 11-15 11-14 11-13	V. Agarwal, JP. Gómez, R. Priestley T. Dimpfl, S. Jank P. Gomber, U. Schweickert, E. Theissen D. Hess, S. Orbe	Title Management Compensation and Market Timing under Portfolio Constraints Can Internet Search Queries Help to Predict Stock Market Volatility? Liquidity Dynamics in an Electronic Open Limit Order Book: An Event Study Approach Irrationality or Efficiency of Macroeconomic Survey Forecasts? Implications from the Anchoring Bias Test
No. 11-16 11-15 11-14 11-13 11-12	V. Agarwal, JP. Gómez, R. Priestley T. Dimpfl, S. Jank P. Gomber, U. Schweickert, E. Theissen D. Hess, S. Orbe D. Hess, P. Immenkötter N. Heinrichs, D. Hess, C. Homburg, M. Lorenz,	Title Management Compensation and Market Timing under Portfolio Constraints Can Internet Search Queries Help to Predict Stock Market Volatility? Liquidity Dynamics in an Electronic Open Limit Order Book: An Event Study Approach Irrationality or Efficiency of Macroeconomic Survey Forecasts? Implications from the Anchoring Bias Test Optimal Leverage, its Benefits, and the Business Cycle Extended Dividend, Cash Flow and Residual Income Valuation
No. 11-16 11-15 11-14 11-13 11-12 11-11	V. Agarwal, JP. Gómez, R. Priestley T. Dimpfl, S. Jank P. Gomber, U. Schweickert, E. Theissen D. Hess, S. Orbe D. Hess, P. Immenkötter N. Heinrichs, D. Hess, C. Homburg, M. Lorenz, S. Sievers A. Kempf, O. Korn,	Management Compensation and Market Timing under Portfolio Constraints Can Internet Search Queries Help to Predict Stock Market Volatility? Liquidity Dynamics in an Electronic Open Limit Order Book: An Event Study Approach Irrationality or Efficiency of Macroeconomic Survey Forecasts? Implications from the Anchoring Bias Test Optimal Leverage, its Benefits, and the Business Cycle Extended Dividend, Cash Flow and Residual Income Valuation Models – Accounting for Deviations from Ideal Conditions

11-07	V. Agarwal, G. D. Gay, L. Ling	Performance inconsistency in mutual funds: An investigation of window-dressing behavior
11-06	N. Hautsch, D. Hess, D. Veredas	The Impact of Macroeconomic News on Quote Adjustments, Noise, and Informational Volatility
11-05	G. Cici	The Prevalence of the Disposition Effect in Mutual Funds' Trades
11-04	S. Jank	Mutual Fund Flows, Expected Returns and the Real Economy
11-03	G.Fellner, E.Theissen	Short Sale Constraints, Divergence of Opinion and Asset Value: Evidence from the Laboratory
11-02	S.Jank	Are There Disadvantaged Clienteles in Mutual Funds?
11-01	V. Agarwal, C. Meneghetti	The Role of Hedge Funds as Primary Lenders
2010		
No.	Author(s)	Title
10-20	G. Cici, S. Gibson, J.J. Merrick Jr.	Missing the Marks? Dispersion in Corporate Bond Valuations Across Mutual Funds
10-19	J. Hengelbrock, E. Theissen, C. Westheide	Market Response to Investor Sentiment
10-18	G. Cici, S. Gibson	The Performance of Corporate-Bond Mutual Funds: Evidence Based on Security-Level Holdings
10-17	D. Hess, D. Kreutzmann, O. Pucker	Projected Earnings Accuracy and the Profitability of Stock Recommendations
10-16	S. Jank, M. Wedow	Sturm und Drang in Money Market Funds: When Money Market Funds Cease to Be Narrow
10-15	G. Cici, A. Kempf, A. Puetz	The Valuation of Hedge Funds' Equity Positions
10-14	J. Grammig, S. Jank	Creative Destruction and Asset Prices
10-13	S. Jank, M. Wedow	Purchase and Redemption Decisions of Mutual Fund Investors and the Role of Fund Families
10-12	S. Artmann, P. Finter, A. Kempf, S. Koch, E. Theissen	The Cross-Section of German Stock Returns: New Data and New Evidence
10-11	M. Chesney, A. Kempf	The Value of Tradeability
10-10	S. Frey, P. Herbst	The Influence of Buy-side Analysts on Mutual Fund Trading
10-09	V. Agarwal, W. Jiang, Y. Tang, B. Yang	Uncovering Hedge Fund Skill from the Portfolio Holdings They Hide
10-08	V. Agarwal, V. Fos, W. Jiang	Inferring Reporting Biases in Hedge Fund Databases from Hedge Fund Equity Holdings
10-07	V. Agarwal, G. Bakshi, J. Huij	Do Higher-Moment Equity Risks Explain Hedge Fund Returns?
10-06	J. Grammig, F. J. Peter	Tell-Tale Tails
10-05	K. Drachter, A. Kempf	Höhe, Struktur und Determinanten der Managervergütung- Eine Analyse der Fondsbranche in Deutschland
10-04	J. Fang, A. Kempf, M. Trapp	Fund Manager Allocation

10-03	P. Finter, A. Niessen- Ruenzi, S. Ruenzi	The Impact of Investor Sentiment on the German Stock Market
10-02	D. Hunter, E. Kandel, S. Kandel, R. Wermers	Mutual Fund Performance Evaluation with Active Peer Benchmarks
10-01	S. Artmann, P. Finter, A. Kempf	Determinants of Expected Stock Returns: Large Sample Evidence from the German Market
2009		
No.	Author(s)	Title
09-17	E. Theissen	Price Discovery in Spot and Futures Markets: A Reconsideration
09-16	М. Тгарр	Trading the Bond-CDS Basis – The Role of Credit Risk and Liquidity
09-15	A. Betzer, J. Gider, D.Metzger, E. Theissen	Strategic Trading and Trade Reporting by Corporate Insiders
09-14	A. Kempf, O. Korn, M. Uhrig-Homburg	The Term Structure of Illiquidity Premia
09-13	W. Bühler, M. Trapp	Time-Varying Credit Risk and Liquidity Premia in Bond and CDS Markets
09-12	W. Bühler, M. Trapp	Explaining the Bond-CDS Basis – The Role of Credit Risk and Liquidity
09-11	S. J. Taylor, P. K. Yadav, Y. Zhang	Cross-sectional analysis of risk-neutral skewness
09-10	A. Kempf, C. Merkle, A. Niessen-Ruenzi	Low Risk and High Return – Affective Attitudes and Stock Market Expectations
09-09	V. Fotak, V. Raman, P. K. Yadav	Naked Short Selling: The Emperor`s New Clothes?
09-08	F. Bardong, S.M. Bartram, P.K. Yadav	Informed Trading, Information Asymmetry and Pricing of Information Risk: Empirical Evidence from the NYSE
09-07	S. J. Taylor , P. K. Yadav, Y. Zhang	The information content of implied volatilities and model-free volatility expectations: Evidence from options written on individual stocks
09-06	S. Frey, P. Sandas	The Impact of Iceberg Orders in Limit Order Books
09-05	H. Beltran-Lopez, P. Giot, J. Grammig	Commonalities in the Order Book
09-04	J. Fang, S. Ruenzi	Rapid Trading bei deutschen Aktienfonds: Evidenz aus einer großen deutschen Fondsgesellschaft
09-03	A. Banegas, B. Gillen, A. Timmermann, R. Wermers	The Cross-Section of Conditional Mutual Fund Performance in European Stock Markets
09-02	J. Grammig, A. Schrimpf, M. Schuppli	Long-Horizon Consumption Risk and the Cross-Section of Returns: New Tests and International Evidence
09-01	O. Korn, P. Koziol	The Term Structure of Currency Hedge Ratios

2008

07-06

V. Agarwal, L. Wang

2000		
No.	Author(s)	Title
08-12	U. Bonenkamp, C. Homburg, A. Kempf	Fundamental Information in Technical Trading Strategies
08-11	O. Korn	Risk Management with Default-risky Forwards
08-10	J. Grammig, F.J. Peter	International Price Discovery in the Presence of Market Microstructure Effects
08-09	C. M. Kuhnen, A. Niessen	Public Opinion and Executive Compensation
08-08	A. Pütz, S. Ruenzi	Overconfidence among Professional Investors: Evidence from Mutual Fund Managers
08-07	P. Osthoff	What matters to SRI investors?
08-06	A. Betzer, E. Theissen	Sooner Or Later: Delays in Trade Reporting by Corporate Insiders
08-05	P. Linge, E. Theissen	Determinanten der Aktionärspräsenz auf Hauptversammlungen deutscher Aktiengesellschaften
08-04	N. Hautsch, D. Hess, C. Müller	Price Adjustment to News with Uncertain Precision
08-03	D. Hess, H. Huang, A. Niessen	How Do Commodity Futures Respond to Macroeconomic News?
08-02	R. Chakrabarti, W. Megginson, P. Yadav	Corporate Governance in India
08-01	C. Andres, E. Theissen	Setting a Fox to Keep the Geese - Does the Comply-or-Explain Principle Work?
2007		
No.	Author(s)	Title
07-16	M. Bär, A. Niessen, S. Ruenzi	The Impact of Work Group Diversity on Performance: Large Sample Evidence from the Mutual Fund Industry
07-15	A. Niessen, S. Ruenzi	Political Connectedness and Firm Performance: Evidence From Germany
07-14	O. Korn	Hedging Price Risk when Payment Dates are Uncertain
07-13	A. Kempf, P. Osthoff	SRI Funds: Nomen est Omen
07-12	J. Grammig, E. Theissen, O. Wuensche	Time and Price Impact of a Trade: A Structural Approach
07-11	V. Agarwal, J. R. Kale	On the Relative Performance of Multi-Strategy and Funds of Hedge Funds
07-10	M. Kasch-Haroutounian, E. Theissen	Competition Between Exchanges: Euronext versus Xetra
07-09	V. Agarwal, N. D. Daniel, N. Y. Naik	Do hedge funds manage their reported returns?
07-08	N. C. Brown, K. D. Wei, R. Wermers	Analyst Recommendations, Mutual Fund Herding, and Overreaction in Stock Prices
07-07	A. Betzer, E. Theissen	Insider Trading and Corporate Governance: The Case of Germany
		The ease of Commany

Transaction Costs and Value Premium

07-05	J. Grammig, A. Schrimpf	Asset Pricing with a Reference Level of Consumption: New Evidence from the Cross-Section of Stock Returns
07-04	V. Agarwal, N.M. Boyson, N.Y. Naik	Hedge Funds for retail investors? An examination of hedged mutual funds
07-03	D. Hess, A. Niessen	The Early News Catches the Attention: On the Relative Price Impact of Similar Economic Indicators
07-02	A. Kempf, S. Ruenzi, T. Thiele	Employment Risk, Compensation Incentives and Managerial Risk Taking - Evidence from the Mutual Fund Industry -
07-01	M. Hagemeister, A. Kempf	CAPM und erwartete Renditen: Eine Untersuchung auf Basis der Erwartung von Marktteilnehmern
2006		
No.	Author(s)	Title
06-13	S. Čeljo-Hörhager, A. Niessen	How do Self-fulfilling Prophecies affect Financial Ratings? - An experimental study
06-12	R. Wermers, Y. Wu, J. Zechner	Portfolio Performance, Discount Dynamics, and the Turnover of Closed-End Fund Managers
06-11	U. v. Lilienfeld-Toal, S. Ruenzi	Why Managers Hold Shares of Their Firm: An Empirical Analysis
06-10	A. Kempf, P. Osthoff	The Effect of Socially Responsible Investing on Portfolio Performance
06-09	R. Wermers, T. Yao, J. Zhao	Extracting Stock Selection Information from Mutual Fund holdings: An Efficient Aggregation Approach
06-08	M. Hoffmann, B. Kempa	The Poole Analysis in the New Open Economy Macroeconomic Framework
06-07	K. Drachter, A. Kempf, M. Wagner	Decision Processes in German Mutual Fund Companies: Evidence from a Telephone Survey
06-06	J.P. Krahnen, F.A. Schmid, E. Theissen	Investment Performance and Market Share: A Study of the German Mutual Fund Industry
06-05	S. Ber, S. Ruenzi	On the Usability of Synthetic Measures of Mutual Fund Net- Flows
06-04	A. Kempf, D. Mayston	Liquidity Commonality Beyond Best Prices
06-03	O. Korn, C. Koziol	Bond Portfolio Optimization: A Risk-Return Approach
06-02	O. Scaillet, L. Barras, R. Wermers	False Discoveries in Mutual Fund Performance: Measuring Luck in Estimated Alphas
06-01	A. Niessen, S. Ruenzi	Sex Matters: Gender Differences in a Professional Setting
2005		
No.	Author(s)	Title
05-16	E. Theissen	An Analysis of Private Investors' Stock Market Return Forecasts
05-15	T. Foucault, S. Moinas, E. Theissen	Does Anonymity Matter in Electronic Limit Order Markets
05-14	R. Kosowski, A. Timmermann, R. Wermers, H. White	Can Mutual Fund "Stars" Really Pick Stocks? New Evidence from a Bootstrap Analysis

05-13	D. Avramov, R. Wermers	Investing in Mutual Funds when Returns are Predictable
05-12	K. Griese, A. Kempf	Liquiditätsdynamik am deutschen Aktienmarkt
05-11	S. Ber, A. Kempf, S. Ruenzi	Determinanten der Mittelzuflüsse bei deutschen Aktienfonds
05-10	M. Bär, A. Kempf, S. Ruenzi	Is a Team Different From the Sum of Its Parts? Evidence from Mutual Fund Managers
05-09	M. Hoffmann	Saving, Investment and the Net Foreign Asset Position
05-08	S. Ruenzi	Mutual Fund Growth in Standard and Specialist Market Segments
05-07	A. Kempf, S. Ruenzi	Status Quo Bias and the Number of Alternatives - An Empirical Illustration from the Mutual Fund Industry
05-06	J. Grammig, E. Theissen	Is Best Really Better? Internalization of Orders in an Open Limit Order Book
05-05	H. Beltran-Lopez, J. Grammig, A.J. Menkveld	Limit order books and trade informativeness
05-04	M. Hoffmann	Compensating Wages under different Exchange rate Regimes
05-03	M. Hoffmann	Fixed versus Flexible Exchange Rates: Evidence from Developing Countries
05-02	A. Kempf, C. Memmel	Estimating the Global Minimum Variance Portfolio
05-01	S. Frey, J. Grammig	Liquidity supply and adverse selection in a pure limit order book market
2004		
2004 No.	Author(s)	Title
	Author(s) N. Hautsch, D. Hess	Title Bayesian Learning in Financial Markets – Testing for the Relevance of Information Precision in Price Discovery
No.	· ·	Bayesian Learning in Financial Markets – Testing for the
No. 04-10	N. Hautsch, D. Hess	Bayesian Learning in Financial Markets – Testing for the Relevance of Information Precision in Price Discovery Portfolio Disclosure, Portfolio Selection and Mutual Fund
No. 04-10 04-09	N. Hautsch, D. Hess A. Kempf, K. Kreuzberg N.F. Carline, S.C. Linn,	Bayesian Learning in Financial Markets – Testing for the Relevance of Information Precision in Price Discovery Portfolio Disclosure, Portfolio Selection and Mutual Fund Performance Evaluation Operating performance changes associated with corporate
No. 04-10 04-09 04-08	N. Hautsch, D. Hess A. Kempf, K. Kreuzberg N.F. Carline, S.C. Linn, P.K. Yadav J.J. Merrick, Jr., N.Y. Naik,	Bayesian Learning in Financial Markets – Testing for the Relevance of Information Precision in Price Discovery Portfolio Disclosure, Portfolio Selection and Mutual Fund Performance Evaluation Operating performance changes associated with corporate mergers and the role of corporate governance Strategic Trading Behaviour and Price Distortion in a
No. 04-10 04-09 04-08 04-07	N. Hautsch, D. Hess A. Kempf, K. Kreuzberg N.F. Carline, S.C. Linn, P.K. Yadav J.J. Merrick, Jr., N.Y. Naik, P.K. Yadav	Bayesian Learning in Financial Markets – Testing for the Relevance of Information Precision in Price Discovery Portfolio Disclosure, Portfolio Selection and Mutual Fund Performance Evaluation Operating performance changes associated with corporate mergers and the role of corporate governance Strategic Trading Behaviour and Price Distortion in a Manipulated Market: Anatomy of a Squeeze Trading Costs of Public Investors with Obligatory and
No. 04-10 04-09 04-08 04-07 04-06	N. Hautsch, D. Hess A. Kempf, K. Kreuzberg N.F. Carline, S.C. Linn, P.K. Yadav J.J. Merrick, Jr., N.Y. Naik, P.K. Yadav N.Y. Naik, P.K. Yadav	Bayesian Learning in Financial Markets – Testing for the Relevance of Information Precision in Price Discovery Portfolio Disclosure, Portfolio Selection and Mutual Fund Performance Evaluation Operating performance changes associated with corporate mergers and the role of corporate governance Strategic Trading Behaviour and Price Distortion in a Manipulated Market: Anatomy of a Squeeze Trading Costs of Public Investors with Obligatory and Voluntary Market-Making: Evidence from Market Reforms Family Matters: Rankings Within Fund Families and
No. 04-10 04-09 04-08 04-07 04-06 04-05	N. Hautsch, D. Hess A. Kempf, K. Kreuzberg N.F. Carline, S.C. Linn, P.K. Yadav J.J. Merrick, Jr., N.Y. Naik, P.K. Yadav N.Y. Naik, P.K. Yadav A. Kempf, S. Ruenzi V. Agarwal, N.D. Daniel,	Bayesian Learning in Financial Markets – Testing for the Relevance of Information Precision in Price Discovery Portfolio Disclosure, Portfolio Selection and Mutual Fund Performance Evaluation Operating performance changes associated with corporate mergers and the role of corporate governance Strategic Trading Behaviour and Price Distortion in a Manipulated Market: Anatomy of a Squeeze Trading Costs of Public Investors with Obligatory and Voluntary Market-Making: Evidence from Market Reforms Family Matters: Rankings Within Fund Families and Fund Inflows Role of Managerial Incentives and Discretion in Hedge Fund
No. 04-10 04-09 04-08 04-07 04-06 04-05 04-04	N. Hautsch, D. Hess A. Kempf, K. Kreuzberg N.F. Carline, S.C. Linn, P.K. Yadav J.J. Merrick, Jr., N.Y. Naik, P.K. Yadav N.Y. Naik, P.K. Yadav A. Kempf, S. Ruenzi V. Agarwal, N.D. Daniel, N.Y. Naik V. Agarwal, W.H. Fung,	Bayesian Learning in Financial Markets – Testing for the Relevance of Information Precision in Price Discovery Portfolio Disclosure, Portfolio Selection and Mutual Fund Performance Evaluation Operating performance changes associated with corporate mergers and the role of corporate governance Strategic Trading Behaviour and Price Distortion in a Manipulated Market: Anatomy of a Squeeze Trading Costs of Public Investors with Obligatory and Voluntary Market-Making: Evidence from Market Reforms Family Matters: Rankings Within Fund Families and Fund Inflows Role of Managerial Incentives and Discretion in Hedge Fund Performance Risk and Return in Convertible Arbitrage:

